# HONORING CULTURE WHILE CHALLENGING PATRIARCHY: FEMINISTS REINTERPRETATION AND REFOMATION OF SHIMGILINA CEREMONY IN ADDIS ABABA

By Kalkidan Negatu

Submitted to Central European University - Private University

Department of Gender Studies

In partial fulfilment of the requirements for the degree Master of Arts in Gender Studies

Supervisor: Professor Eva Fodor

Second Reader: Associate Professor Hadley Z. Renkin

Vienna, Austria 2025

#### Author's declaration

I hereby declare that this thesis is the result of original research; it contains no materials accepted for any other degree in any other institution and no materials previously written and/or published by another person, except where appropriate acknowledgment is made in the form of bibliographical reference.

I further declare that the following word count for this thesis are accurate:

Body of thesis (all chapters excluding notes, references, appendices, etc.):23,572 words Entire manuscript: 25,511 words

Signed Kalkidan Negatu

(Signature appears on the hard copy submitted to the library)

### **Copyright Notice**

Copyright © Kalkidan Negatu, 2025. Honoring Culture While Challenging Patriarchy: Feminists Reinterpretation and Reformation of Shimgilina Ceremony in Addis Ababa - This work is licensed under <u>Creative Commons Attribution-NonCommercial-NoDerivatives (CC BY-NC-ND) 4.0 International</u> license.

#### **Abstract**

This thesis explores how urban, middle-class Ethiopians who identify as feminists engage with *Shimgilina*—a pre-wedding ceremony that, while deeply rooted in patriarchal tradition, remains central to Ethiopian marriage practices. The ritual often reinforces male dominance through male-only negotiations and, in rural contexts, is closely linked to child marriage. Yet, *Shimgilina* is also seen as a vital expression of cultural identity, a means of preserving heritage, and a way of fostering bonds between families. Amid these tensions, many feminist women and men still choose to include the ceremony in their marriage process. Given their relative privilege and autonomy, this participation is unlikely to stem from coercion. Instead, it prompts a more complex question: what forms of agency are being exercised within this deeply gendered cultural space?

Using qualitative research methods, I conducted semi-structured interviews with ten self-identified feminists—nine women and one man—based in Addis Ababa. While participants shared similar social locations, they held divergent perspectives and strategies for navigating *Shimgilina*.

In analyzing these experiences, I drew on four theoretical frameworks—intersectionality, patriarchal bargaining, Nego-feminism, and Saba Mahmood's rethinking of agency—to trace how feminist agency is expressed within the Shimgilina ceremony. Two distinct patterns emerged. The first group expressed a strong sense of autonomy prior to the ceremony, particularly through independently choosing their partners. Influenced by this autonomy and by cultural expectations they embraced, they did not perceive Shimgilina as a site of struggle. Instead, they viewed it as a symbolic gesture—one that legitimized their personal choices in the eyes of their families, strengthened new relational bonds, and affirmed their cultural belonging.

In contrast, the second group did not articulate the same level of autonomy beforehand—not because it was absent, but because it was not felt or foregrounded in their narratives. For them, the ceremony itself became the key site through which to assert agency. Drawing on their feminist consciousness, they intervened more visibly—reshaping the ritual by including maternal figures, pushing for gender-inclusive representation, or challenging patriarchal norms embedded in the process.

Overall, this research reveals that agency within Shimgilina is shaped by culture, power, and a fluid sense of autonomy. Even within a relatively homogeneous group, expressions of feminist agency vary significantly, making the practice deeply contextual, yet plural and internally diverse.

#### Acknowledgement

I would like to express my deepest gratitude to my family and friends who supported me throughout this journey. To my mother, who instilled in me the value of education and perseverance. For my brothers, for always believing in me and hyping me up. For my boyfriend, whose emotional support helped me carry the weight of the process. For my dearest friend, Arsema Getachew, for always cheering me on.

I would also like to thank my supervisor and second reader for their guidance and encouragement, and for pushing me to make the most of this educational experience and thesis journey.

## **Table of Contents**

Abstractiii										
Α	Acknowledgementv									
1 Introduction										
	1.1	Feminist Engagement with Tradition	1							
	1.2	Significance of Shimgilina and its gendered roots	2							
2	Ethi	opia in Context	6							
	2.1	History, Class, Gender, and Residence	6							
3	Lite	rature Review	.10							
	3.1	Introduction	10							
	3.2	Marriage and its meaning	.12							
	3.3	Wedding ceremonies as sites of gendered performance	14							
	3.4 Comm	Gendered Agency in Marriage Proposals: Absent, Constrained, or/Bothnercialized?	20							
	3.5	Shifting tradition: Between preservation and reinterpretation	.21							
4	Con	nceptual Framework	. 24							
	4.1	Intersectionality	24							
	4.2	Patriarchal Bargaining	. 25							
	4.3	Nego Feminism	. 28							
	4.4	Agency Conceptualization	30							
5	Met	hodology	.34							
	5.1	Data Collection	.34							
	5.2	Data Analysis	36							
	5.3	Ethical Considerations	37							
	5.4	Positionality	37							
	5.5	Limitations	.40							
	5.6	Methodological tools	.40							
	5.6.	1 Feminist Methodology	40							

	5.6.2	Intersectionality as a Method	41
	5.6.2.	1 Gender	43
	5.6.2.2	Nationality	44
	5.6.2.3	3 Capital City Residence	46
	5.6.2.4	4 Class	47
	5.6.2.	5 Age	48
	5.6.3	Feminism in Context: Intersecting Realities of Ethiopian Feminists	48
6	Navigati	ing Tradition: The Emergence of Two Feminist Approaches to Shimgi	ilina52
6	3.1 Reir	nterprets: Making Feminist Sense of Tradition	53
	6.1.1	Feminist Critique by Reinterpreters	53
	6.1.2	Social Bonds and Cultural Legitimacy: Rewriting the Rules of Shimgilin	a56
	6.1.3	From Oppression to Expression: Fighting Patriarchy with Autonomy	59
	6.1.4	Navigating Feminist Agency through Cultural Attachment	67
6	6.2 Refe	ormers: Feminist Shifts in Ceremony	70
	6.2.1	Feminist Critique by Reformers	70
	6.2.2	Shimgilina as Protection, Belonging, and Identity	71
	6.2.3	Shaping the Ritual: Autonomy Through Cultural Influence	75
	6.2.4	Conclusion	85
7	Conclus	sion	87
Bib	liography	1	91

# List of figures

г' 1	M	CE41	1 1/4 1	11 '			_
Figure I	: мар о	i Ethiopia	and its nei	gnboring	g countries	(	3

#### 1 Introduction

#### 1.1 Feminist Engagement with Tradition

"They were told to come early in the morning — by 5 a.m. — so that's when they showed up.

From my side, my uncles and the other male elders had already been asked properly if they would serve as my *shimagile*(elder), and they took it seriously. It's a role of honor. Even my father, when he's asked to represent someone else, always says, 'I was one of her *shimagile*,' with pride.

When the time came, things started to feel real. The men in my family — my uncles, male cousins, and the elders — began to gather. I quietly slipped away to my room, because that's what you're supposed to do. When they said it was almost time, I knew I had to hide.

Then the discussion began. They were figuring out what they would ask — and more importantly, who would speak. It had to be someone older, someone respected, someone who could speak well on behalf of our family.

Everyone dressed so nicely, from both sides. You can't just show up to something like this in a T-shirt and jeans. It's a big day.

When the elders from his side arrived, they were late. My side of the family teased them: 'How can we give you our daughter if you can't even show up on time?' they said.

Then the shimgilaes (elders) from the other side walked in— all men, dressed in tuxedos, standing motionless.

When they were invited to sit down, they said, politely, "Not yet — we have things to ask first." And so, they began. They told my family that their son is interested in marrying me. Then my family started asking questions:

'Who is he?' 'Who are his parents?' 'How is he planning to provide for her?' 'What property does he have?' They even asked about his behavior.

Since we're not from the same ethnic group, they didn't go too deep into lineage — but if we had been, I think they would have.

The elders from his side responded confidently. They said he was educated, had a stable job, and seemed responsible. 'He's a good guy,' they said. 'They'll support each other.'

Then my parents added, 'She's educated, too.'

Finally, after some back and forth, someone from my side asked, 'If we give you our daughter, when will you arrange the marriage?'

And his side said, 'They will decide together.'

Only then were they given permission for their son to marry me, and they were invited to sit. Food was brought out, and the atmosphere began to soften.

I was still up in my room, hiding. People came in and out, whispering updates, telling me what was happening. They were taking pictures of me — I was the center of attention, even though I wasn't in the room.

Eventually, my mother and grandmother came out to be introduced, and then the women in my family began to serve the food.

That's how it happened." (Tsige, 28)

This vignette, drawn from a female participant's experience, outlines the core elements of a typical Shimgilina ceremony. While details vary, consistent features include the groom sending male elders, their central role in negotiations, the importance of familial approval, and symbolic acts of respect, such as elders refusing to sit before acceptance.

Although the ceremony reinforces male authority through practices like male-only representation, many feminist women and men in Addis Ababa still choose to include it in their marriage process. This study investigates why, focusing on how young, urban, middle-class both Ethiopians, like Tsige, negotiate and express agency within this gendered ritual.

#### 1.2 Significance of Shimgilina and its gendered roots

To understand the significance of Shimgilina, it must be placed within its broader socio-cultural context. In many African societies, including Ethiopia, marriage is viewed less as a private, romantic union and more as a strategic alliance that serves family and community interests, primarily focused on procreation and the continuation of lineage (Dessie and Bekelcha, 2023)As

a result, the initial stage of marriage in Ethiopia emphasizes compatibility—especially in wealth, status, and religion—as families assess each other to uphold social norms and family honor (Mengesha, Deressa, and Imagnu 1996). Across Ethiopia's diverse ethnic groups, the man's family typically initiates the process, followed by negotiations over timing, terms, and the exchange of gifts or bridewealth (Dessie & Bekelcha, 2023).

Despite its cultural significance, Shimgilina has received limited scholarly attention. It is typically mentioned only in passing within broader discussions of marriage and marriage ceremonies, which is framed as an expression of ethnic identity or cultural tradition, as noted in the literature review. As a result, Ethiopian scholars who do address *Shimgilina* tend to present it in a positive light, often emphasizing its role as a pillar of societal stability (Telila,2020) or as a means of cultural continuity(Beyene, 2006).

While marriage and its ceremonies are widely celebrated, rituals such as shimgilina often involve gendered practices that disadvantage women. In many traditions, it is men who initiate the marriage process (Telila, 2020; Beyene, 2006) fathers who arrange the unions (Dessie and Bekelcha, 2023) and male elders or the groom's family, who lead the negotiations (Beyene, 2006). These customs reinforce male authority and decision-making, leaving women largely passive participants in their marital arrangements. Furthermore, women are frequently treated as transactional objects in the form of bridewealth or dowry (Balew Liyew, 2024). Such practices significantly undermine women's agency by excluding them from critical decisions about their futures and delimiting their participation in economic activities.

This practice, deeply intertwined with cultural and ethnic identity, remains highly prevalent. As a result, many girls are married before reaching the age of 18, making Ethiopia one of the top 15

countries globally in terms of child marriage prevalence (Erulkar 2022). To make matters worse, these early marriage practices expose women to serious risks, including marital rape (Erulkar 2013) and heightened vulnerability to intimate partner violence (Abera et al. 2020).

The ceremony both celebrates communal cultural identity and reinforces patriarchal norms that have serious detrimental effects for women. For Ethiopian feminists planning to marry, this creates a complex tension between honoring tradition and challenging gendered structures.

The contradictions within Shimgilina are what initially drew me to this research—especially after attending the ceremonies of two close feminist friends. Their decision to include a ritual deeply rooted in patriarchy raised pressing questions: How do feminist values coexist with cultural practices that still marginalize and disadvantage women? Is participation in such ceremonies an act of agency or conformity? These tensions became the starting point for my inquiry.

This study explores how Ethiopian feminists engage with rituals like Shimgilina, which sit at the intersection of gendered power and cultural identity. I conducted interviews with ten self-identified Ethiopian feminists in Addis Ababa, all of whom are urban, middle-class, and educated who experienced the Shimgilina ceremony at a young age. The group included eight women who held traditional Shimgilina ceremonies, one who adapted the ceremony to her interest, and one feminist man who chose to send elders for his wedding.

Through their stories, I examine how feminism is lived not by rejecting tradition outright, but by critically engaging with it. Drawing on intersectionality, patriarchal bargaining, Saba Mahmood's concept of embodied agency, and Nego-feminism, this study shows how participants actively reinterpret or reshape Shimgilina from within.

The significance of this research lies in its effort to position culture not as an obstacle to feminism, but as a space through which feminist agency can be enacted. In the Ethiopian context—where tradition deeply shapes social life—this study challenges the assumption that feminism must stand apart from or against culture. Instead, it demonstrates how agency can be understood as relational and situated within a web of cultural norms, family expectations, and personal values. By recognizing these women's actions within their broader social context, the study offers a more nuanced, context-sensitive, and generative understanding of feminist agency.

This thesis is organized into seven chapters, each building toward an understanding of how feminist-identified Ethiopians navigate the Shimgilina ceremony and assert their agency. Chapter 2 provides essential socio-cultural context on Ethiopia's history, economy, gender relations, and demographics, situating participants within the broader structures that shape their agency. Chapter 3 outlines the qualitative, feminist methodology—covering data collection, analysis, positionality, and limitations—and establishes the reflexive and intersectional-sensitive approach of the study. Chapter 4 reviews relevant literature, including Ethiopian studies on marriage customs, feminist research on child marriage, and global feminist work on wedding rituals. It highlights the absence of in-depth analysis on Shimgilina and situates the study within academic works that focus on wedding rituals and/or gender. Chapter 5 introduces the theoretical frameworks—intersectionality, patriarchal bargaining, Nego-feminism, and Saba Mahmood's theory of agency—that guide the analysis and support a contextually grounded reading of feminist agency. Chapter 6 presents the core analysis, showing how participants engage with Shimgilina in diverse ways depending on their social positioning and feminist orientation. Finally, Chapter 7 concludes the thesis by summarizing the key findings and reflecting on their broader implications for feminist theory and cultural negotiation in Ethiopia.

#### 2 Ethiopia in Context

#### 2.1 History, Class, Gender, and Residence

This chapter provides a contextual overview of Ethiopia, outlining the socio-cultural, economic, and political dynamics that shape gender relations and feminist negotiation. Ethiopia's anti-colonial history, diverse population, and strong traditions form the backdrop for understanding how Addis Ababa–based feminists engage with culture.



Figure 1: Map of Ethiopia and its neighboring countries

Ethiopia is situated in the Horn of Africa, in the northeastern part of the continent. As a landlocked nation, it shares borders with Sudan to the west, Eritrea and Djibouti to the northeast, Somalia to the east, and Kenya to the south. Its geographic closeness to the Arabian Peninsula has historically drawn the interest of European colonial powers. Nevertheless, Ethiopia has notably resisted colonization, successfully safeguarding both its sovereignty and its vibrant cultural heritage (Pratelli, 2019)—except for a brief Italian occupation between 1936 and 1941(King and Schramme 2019).

This historical reality has shaped a distinctly Ethiopian philosophical outlook—one that values self-respect, national pride, and a deep commitment to safeguarding independence. Rooted in a legacy of resistance to European colonial invasion, this perspective embodies a spirit of sacrifice and sovereignty, often captured in the phrase "ተፈርታ እና ተከብራ የኖረች ሀገር"—a nation both feared and revered. (Hailu 2021).

Despite strong cultural continuity and a shared national identity, Ethiopia is marked by immense internal diversity. With over 80 ethnic groups and multiple religions—Christianity, Islam, Judaism, and indigenous beliefs—the country is often described as a "museum of peoples," reflecting its rich mosaic of identities. The coexistence of unity and diversity in Ethiopia has been a complex and ongoing challenge. Under the imperial regime (pre-1974), national identity was promoted by suppressing ethnic and cultural differences, leading to widespread marginalization. The Derg regime (1974–1991) acknowledged diversity rhetorically but failed to ensure equitable inclusion. The current federal system, introduced by the Federal Democratic Republic of Ethiopia, recognizes ethnic autonomy and self-determination. However, while it aims to protect cultural rights, it has also deepened divisions and contributed to recurrent ethnic tensions and political instability (Adamu 2013).

In terms of population distribution, Ethiopia remains one of the least urbanized countries in the world. Only about 16% of the population lives in urban areas, while the vast majority—around 70%—live in farming communities (King and Schramme 2019). Addis Ababa, often referred to as the "capital of Africa," serves as a political and economic hub, hosting major institutions like the African Union (AU), and United Nations Economic Commission for Africa UNECA. Its

diplomatic status, regional trade networks, and strategic role in air transport position it as a key site for the exchange of people, goods, and ideas—distinguishing it from rural Ethiopia (Gebrekirstoskassa, 2018).

However, despite Addis Ababa's prominence and periods of national economic growth, Ethiopia remains one of the world's poorest countries, with average incomes well below those of many other low-income nations. (Dorosh and Schmidt, 2010).

These structural economic challenges are compounded by deeply rooted gender inequalities. Ethiopia is a deeply patriarchal society, where women are often relegated to subordinate roles, with religion and culture frequently used to justify gender inequality—reinforced by legal and institutional frameworks. This has created significant disparities between men and women in nearly every aspect of life, from household dynamics to state representation. The status of women in Ethiopia remains critically low: they are more likely to live in poverty, have limited access to education, and often bear the responsibility of heading households without sufficient social or economic support. (Cherinet and Mulugeta 2003)

Even in sectors like agriculture, where women contribute significantly, their labor is often unrecognized and undervalued. While they may be central to family and community survival, women generally lack decision-making power in both public and private spheres. Globally, women outnumber men in many populations, but this is not the case in Ethiopia—partly due to the prevalence of harmful traditional practices. These include female genital mutilation (FGM), gender-based discrimination in food allocation, early marriages—sometimes as young as seven or

eight—rape, abduction followed by sexual violence, and other violations that impact women's health, autonomy, and overall well-being (Cherinet and Mulugeta 2003)

.Moreover, women are disappointedly affected by Poverty because they often lack access to financial resources while bearing household responsibilities. In rural areas, men are seen as primary providers, while urban women are overrepresented in low-paying jobs. Women also face longstanding barriers to land ownership and legal rights, with inheritance and property laws historically favoring men (Sande Lie and Mesin, 2018).

#### 3 Literature Review

#### 3.1 Introduction

Marriage ceremonies across cultures are more than celebratory rituals—they are powerful medium in which societal values are performed, contested, and reimagined. Across diverse contexts, wedding rituals often function as mechanisms through which gender roles and power structures are both produced and reinforced. Yet, they are also vibrant expressions of culture and identity—sites where communities affirm who they are, even as they navigate change.

In the Ethiopian context, *Shimgilina*—a pre-wedding marital negotiation typically initiated by the groom's family—is rooted in patriarchal traditions. While men customarily lead the process and women's agency is often absent or constrained, the ceremony also involves a symbolic exchange in which the bride is associated with material transactions, such as the offering of money. Nonetheless, it remains a practice through which culture is preserved, ethnic identity is expressed, and social bonds are formed.

In this study, I aim to examine how self-identified feminists in Addis Ababa navigate the tension between their cultural values and feminist commitments. While *Shimgilina*—a pre-wedding ceremony rooted in patriarchal norms—can reinforce harmful practices and limit women's autonomy, many feminists still choose to include it in their wedding process. By interviewing feminists who consciously opted to uphold this tradition, I explore how they express agency and

negotiate the complexities of honoring cultural identity while remaining committed to feminist ideals.

To help me understand this negotiation or "give and take," I draw on three bodies of literature. First, I engage with descriptive ethnographic accounts —often produced by Ethiopian male scholars using ethnographic and anthropological methods—that explore the historical and cultural significance of the *Shimgilina* ceremony across different ethnic groups in rural parts of Ethiopia.

Second, I turn to studies conducted in Ethiopia, most of which rely on community-based cross-sectional research or meta-analyses focused on rural areas. These works incorporate a gendered analysis by centering women lived experiences, allowing me to engage more critically with the patriarchal power dynamics embedded in such rituals. By examining these perspectives, I aim to better understand the more harmful and oppressive dimensions *Shimgilina* and its impact on Ethiopian women.

Third, I draw on global feminist literature on wedding rituals, which—though primarily focused on Western contexts—offers valuable insights into how gender is constructed and reinforced through tradition. These studies highlight the patriarchal symbolism embedded in ceremonies, as well as other intersecting systems at play, such as racism, capitalism, and class. While set in places like the United Status of America, United Kingdom, and South Africa, this literature informs my analysis of *Shimgilina* in urban Ethiopia, where gender, agency, and negotiation are similarly at play.

#### 3.2 Marriage and its meaning

Each of the three scholarly bodies of literature approaches marriage through a distinct lens, shaping how they interpret the meaning and function of wedding rituals. While the first group tends to romanticize marriage and its ceremonies, another centers women lived experiences, and a third focuses on the symbolic elements of marriage to unpack underlying gender norms.

The descriptive ethnographic accounts frame marriage as a beneficial and highly respected tradition, situating it within a broader narrative that regards marriage as a foundational social institution. Accordingly, marriage is frequently defined as a union between a man and a woman (Kifle 2013; Dessie and Bekelcha, 2023) oriented toward procreation (Dessie and Bekelcha, 2023) and upheld as a cornerstone of societal stability (Telila, 2020). It is also described as essential for cultural continuity (Balew Liyew, 2024).

As an extension of these values, *Shimgilina* and other marriage-related ceremonies are typically portrayed uncritically as rituals that uphold traditional norms and reinforce societal cohesion. In some accounts, *Shimgilina* is even celebrated as a cultural asset with the potential to contribute to tourism and national economic development (Tătar et al, 2024). The ceremony is also framed as a medium for expressing ethnic identity (Beyene, 2006), facilitating bonds between families (Worku Kalore, 2019), and preserving communal customs (Telila, 2020). In certain cases, it is promoted as an intangible cultural heritage to be protected and supported by regional cultural and tourism departments, given its cultural significance and economic value for both the families involved and the broader community (Dessie and Bekelcha, 2023).

Additionally, this conceptualization of marriage as a beneficial and essential social institution tends to focus on documenting—and, in many ways, celebrating—how various ethnic groups in Ethiopia perform the ceremony, rather than critically examining or challenging its underlying assumptions. As a result, these scholarship papers, often focuses on cultural symbolism within marriage rituals. Among the Hadiya, elders bring honey in a clay pot to signify kindness (Worku Kalore, 2019); in Arsi Oromo tradition, a *siinqee* stick is placed at the bride's door to begin courtship (Telila,2020). In the Amhara region, a successful first meeting leads to a formal invitation called *ketero* for further negotiation (Dessie and Bekelcha, 2023.).

On the other hand, scholars who critically engage with the gender component in the ceremony do not consider it "as a way of life", but rather as a key mechanism in perpetuating child marriage among girls. By shifting the analytical lens from the ritual itself to the lived experiences of the girls involved, the language used in the literature changes significantly—those previously referred to as "brides" become "child brides and the institution of "marriage" is reframed as "child marriage" (Abera et al. 2020; John et al. 2019).

For this group, these rituals are understood as arranged marriages initiated by parents, often without the girl's consent (John et al. 2019). These arrangements are often framed as beneficial, either for the family or even for the girl herself. The logic of the arrangement lies in the motivation and desire to strengthen family ties (Abera et al. 2020; Kok et al. 2023), to protect girls from perceived threats such as premarital pregnancy—which could bring shame and dishonor to the family (Kok et al. 2023)—or to shield her from Gender Based Violence (Abdurahman, Assefa, and Berhane 2022). In many cases, those most likely to be married off through these arranged

ceremonies are young girls who remain under the full control of their parents, with little to no say in the decision (Erulkar 2013; Gebeyehu et al. 2023)

Alongside Ethiopian scholarship that either critiques *Shimgilina* as a site of gendered harm or romanticizes it as an expression of ethnic/national identity, global feminist literature offers a different lens—one that focuses on the ritual itself as a performative space where power and gender are enacted. Studies on the Western style "white wedding," reveal how rituals reproduce patriarchal norms through symbolic acts that position women in passive or subordinate roles (Ellingsæter 2023; Froschauer and Durrheim 2019)

#### 3.3 Wedding ceremonies as sites of gendered performance

The three bodies of literature I engage with each approach gendered performance in wedding rituals from distinct perspectives. Ethnographic and anthropological studies tend to describe visible roles—what men and women do during the ceremony. By contrast, scholars that problematize the ritual, focus on underlying dynamics such as age disparities and the broader consequences of these cultural practices for women. The third group, rooted in global feminist scholarship, analyzes symbolic performances tied to capitalism, gender, race, and other intersecting structures of oppression.

In the descriptive literature, gender roles appear in various ways across different ethnic groups—particularly in accounts that highlight how the marriage process is typically initiated by the man or his family. Often, the groom provides money, informs his father, or has male elders sent as

representatives on his behalf. Again, these observations remain largely descriptive and normalize gendered power dynamics.

In the Arsi Oromo context, for example, it is described the young man identifies the girl he wants to marry and informs his father, who then contacts someone from the bride's family (Telila, 2020). In other cases, such as in east Gojjam, when there is a close relationship between families, a betrothal marriage may be arranged directly by the fathers of the bride and groom—a practice described as both common and respected (Dessie and Bekelcha, 2023). It is also often the groom's parents who are responsible for selecting a "suitable bride" for their son (Augustyniak 2009;Beyene,2006). Among the Hadiya, although the full arrangement process is not always clear, three men are typically sent to the bride's home to formally request the union (Worku Kalore, 2019). Even in historical accounts, such as among the Afar Islamic community in the 10th century, it is the man who approaches the bride's father with special gifts to initiate the request (Augustyniak 2009)

Across many ethnographic accounts, the premarital negotiation phase is presented as a central moment for the exchange of bride wealth and dowries (Balew Liyew 2024; Tătar et al. 2024) However, these practices are often described as routine or culturally normative, with little critical attention given to their implications or consequences.

In many cases, if the bride's parents are open to marriage, they receive money or goods as part of the agreement. Among the Hadiya, for example, the groom's family gives 200 euros to the bride's father and 100 euros to her mother (Worku Kalore, 2019). In Amhara marriage practices, bride

wealth is typically paid in either cash or kind. It is usually the boy's father who provides the payment to the bride and her family. In return, the bride's father gives an ox or sheep to the groom's father. The bride herself is often gifted with gold. These exchanges are framed to stabilize and secure the marriage arrangement (Dessie and Bekelcha, 2023). In the Gidda Oromo tradition, a specific form of marriage arrangement known as *Hawi* also involves gendered economic exchange. In this case, the boy asks his father to negotiate on his behalf. If the negotiation is successful, the father provides his son with money to buy clothes, shoes, and an umbrella for the bride (Beyene, 2006).

Women are largely absent from the narratives presented in much of the ethnographic literature on *Shimgilina*. When these ceremonies are described, women are either overlooked entirely or presumed to have consented to the marriage. The accounts often center the man's agency: he decides to marry, informs his father, and male elders are sent as representatives. In contrast, women are neither visibly present in the ceremony nor granted decision-making roles in the literature. In many cases, it is unclear whether women are allowed to accept or refuse the marriage proposal. In one account, a girl is described as "encouraging," with her best friends acting as intermediaries to communicate her feelings to the groom's family (Telila, 2020). Even in the exchange of wealth, men dominate the process, and it reinforces the idea that women are commodities that are being bought from their fathers. Moreover, even in the cases that they receive money, the bride or even her mother receives significantly less. As an example, I gave earlier, when the bride's father receives 200 euros, the mother 100 euros, while the bride herself receives nothing(Worku Kalore, 2019). Similarly, in Amhara traditions, the groom's family provides livestock to the bride's father, while the mother is given only clothing (Dessie and Bekelcha, 2023).

Scholars who critically engage with gender component in the Ethiopian context of *Shimgilina* highlight power imbalances often overlooked in descriptive accounts. This literature interrogates decision-making dynamics, where the man's role in choosing and paying for a bride is closely tied to age and authority. They highlight that such arrangements are widespread in Ethiopia: 58% of women marry before the age of 18, compared to only 9% of men. In extreme cases, particularly in the Amhara region—marriage promises are reportedly made even before a girl is born(A. M. Dessie et al. 2023).

A key concern is that man's older age and financial control create a harmful power imbalance, limiting the younger bride's autonomy and reinforcing a hierarchical relationship (John et al. 2019). Moreover, age is closely tied to age. They show that girls who marry at or after age 18 are significantly more likely to participate in household decision-making, while those married earlier often have little to no agency over their lives (Abera et al. 2020). In some instances, girls married before age 15 are not even aware of the marriage until the arrangements are already underway (Eruklar, 2013). These early and often one-sided unions strip girls of power from the outset, placing them in highly vulnerable positions (Abera et al. 2020; John et al. 2019).

The literature highlights that the consequences of early marriage extend far beyond the wedding ceremony, with deeply harmful and lasting effects on girls and women. The consequences of early marriage are extensive and profoundly damaging. Child brides are often subjected to marital rape during their first sexual experience (Erulkar 2013) and face a significantly higher risk of enduring spousal abuse and domestic violence throughout their marriages (Abera et al. 2020). Moreover, the lack of agency not to choose a partner initiated contributes to patterns of violence and

undermines trust, intimacy, communication, and the overall quality of the relationship (John et al. 2019).

Global feminist scholarship examines pre-wedding and wedding ceremonies by analyzing the symbolic systems, language, and gendered performances that are woven into the rituals themselves. The engagement ring is viewed as a symbol indicating that a woman is "taken" (Baker and Elizabeth, 2013), or even as a form of partial payment (Froschauer and Durrheim 2019). Ritual moments such as the bride being "given away" symbolize the history of women being married off without their explicit consent (Froschauer and Durrheim, 2019), the groom "asking her father" for permission gives her father authority over her, or the entitlement to "kiss the bride," which reflect institutionalized gendered power imbalances(Fairchild 2014). The white wedding dress symbolizes purity and virginity, reinforcing cultural ideals of feminine innocence (Froschauer and Durrheim, 2019; Fairchild 2014).

Further critiques focus on the wedding industry as a powerful site where patriarchy and capitalism intersect. The role of a woman in weddings is shaped to serve the capitalist system. She is told to be picture-perfect, so the bride goes to great lengths, engaging in unhealthy diets, costly beauty treatments, and exhaustive self-preparation, to embody the ideal (Dewandaru and Triastuti 2021). Popular culture portrays wedding as the pinnacle of a woman's life and her opportunity to become a temporary celebrity, encouraging excessive consumption (Baker and Elizabeth, 2013). These dynamics are driven the "wedding industrial complex," where gendered expectations are not only reproduced but monetized through the collaboration of media, tradition, and market forces (Dewandaru and Triastuti, 2021)

Moreover, Bridal media further promotes this narrative by placing the responsibility of wedding planning and emotional labor squarely on women, framing unpaid care work as part of the bride's role (Carter 2022). Scholars problematize this dynamic by pointing out that men's contributions, such as proposing, often require significantly less effort and planning. In fact, the emotional labor expected of a proposal is frequently minimal, likened to simply saying "Happy Birthday". Meanwhile, women are expected to plan the wedding, organize events like prom, and manage family gatherings. This reflects a broader patriarchal division of labor, often described as the split between "frontstage" and "backstage" performances. While grooms may appear to take initiative, they are often following their fiancée's detailed instructions—maintaining the illusion of male leadership in line with social expectations. Couples frequently navigate these roles in ways that feel authentic to them, while still conforming to societal norms (Schweingruber, Anahita, and Berns, 2004)

Beyond gender, this body of literature also addresses other intersecting forms of oppression, such as class and race. It highlights how white weddings often reproduce idealized notions of femininity that are deeply racialized and classed. The "ideal bride" is frequently portrayed as white, middle-class, and hyper-feminine. Cultural narratives—such as the "bridezilla" trope—serve to discipline women who deviate from these expectations, especially working-class and non-white brides, who are often depicted as excessive, unruly, or insufficiently feminine (Carter, 2022).

# 3.4 Gendered Agency in Marriage Proposals: Absent, Constrained, or/Both Commercialized?

In the descriptive ethnographic accounts, it often remains unclear whether women could agree to or refuse a marriage proposal. Only in one instance, the girl is described as "encouraging," with her best friends communicating her wishes to the groom's side (Telila, 2020). In the Ethiopian scholarship that problematizes gender, women's agency is portrayed as either absent or severely constrained. Parental authority dominates the decision-making process (Gebeyehu et al. 2023), with marriage decisions typically made by the parents rather than the girl herself (John et al. 2019). In some cases, girls younger than 15 may not even be aware that they are getting married until the process is already underway (Erulkar 2013).

In global feminist discourse, there is a more nuanced conceptualization of agency, particularly in relation to marriage proposals. Despite significant advancements in gender equality—such as increased female participation in the workforce, gender-neutral legislation, and evolving norms around domestic labor—certain patriarchal rituals, like the tradition of men proposing, remain largely unquestioned by women (Ellingsæter 2023). Female-initiated proposals are often seen as less legitimate, and women themselves sometimes pressure their peers to get married, even in contexts where feminist ideas and systemic gender progress are widely embraced (Baker and Elizabeth, 2013).

This apparent contradiction, scholars from this group argue, is partly due to the privatization of choice: acts like accepting a proposal or planning a wedding are framed as personal decisions, thereby shielding them from feminist critique (Ellingsæter 2023). Moreover, the rise of "choice

feminism" supports this framework, suggesting that any decision a woman makes—whether to wear makeup, stay home, or plan an extravagant wedding—is valid as long as it is freely chosen. As some scholars in this group contend, the emphasis on the individual often suppresses political debate. It is merely seen as critiquing a choice is seen as attacking a woman's agency (Carter 2022). Other scholars have this contradictory phenomenon, as an influence of the wedding industry, which markets femininity as a commodity to fulfill a woman's desire to belong to society. This creates an emotional relationship between women as consumers and the rituals of marriage. As such, personal agency becomes entangled with market-driven ideals, making feminist resistance more complex in the symbolic world of weddings (Dewandaru and Triastuti, 2021).

#### 3.5 Shifting tradition: Between preservation and reinterpretation

In the descriptive ethnographic accounts, *Shimgilina* is portrayed as a fixed and unchanging tradition, simply a part of life, leaving little room for negotiation or reinterpretation. It is presented as a rigid expression of identity, rather than something that can be changed. Similarly, the other group within Ethiopian scholarship does not critically examine the ritual itself or consider it as a potential site for negotiation. Instead, proposed solutions tend to focus on protecting girls through structural interventions, such as implementing community-based programs to keep girls in school (Erulkar 2013) and increasing women's decision-making power in rural areas (Gebeyehu et al. 2023).

Global feminist literature often frames tradition as flexible, layered with contradictions, and open to negotiation. While these scholars reflect this view through varied frameworks, they consistently emphasize how individuals and couples do not simply follow fixed cultural paths. Instead, they navigate between personal values and social expectations, adapting rituals in ways that resonate with their lived realities. For example, when traditional gender norms—such as the bride vowing obedience—clash with egalitarian ideals, couples may revise their vows to reflect equality while still preserving other patriarchal elements, like the father giving away the bride (Fairchild 2014).

Moreover, scholars in this group critique simplistic classifications such as "resisters" or "maintainers," arguing that they fail to capture the nuanced ways people engage with tradition. Within a single relationship, individuals may hold contradictory views or selectively adapt different aspects of various rituals (Ellingsæter 2023). Tradition, from this perspective, is malleable—constantly reshaped through reinterpretation and everyday practice. Concepts like detraditionalization and bricolage are used to describe how people integrate modern values into traditional forms, not by rejecting them entirely but by reworking them to fit new meanings (Carter and Duncan, 2017).

While each body of literature offers valuable insights, significant gaps remain in understanding the gendered dimensions of marriage rituals like *Shimgilina* in Ethiopia. Ethnographic and anthropological accounts tend to be descriptive, often portraying culture as fixed and focusing largely on rural settings. These works rarely engage with questions of power, gender dynamics, or the possibility of change within tradition. Critical gender scholarship in Ethiopia, on the other hand, offers important analyses of patriarchy and the structural oppression women face—especially in

relation to child marriage and rural life. However, this body of work often overlooks the ritual itself as a potential site of negotiation and meaning making, even in contexts where the woman has given consent and is older than eighteen years old, educated and living in an urban setting. Meanwhile, global feminist literature provides robust frameworks for analyzing agency, symbolism, and negotiation within wedding rituals, but it lacks attention to the Ethiopian context, particularly the complex interplay of class, culture, and race.

While the reviewed literature provides important insights, it remains limited in capturing the nuanced dynamics of gendered agency within the Shimgilina ritual. Descriptive ethnographies often portray the ceremony as a static cultural tradition, lacking attention to the evolving power relations it entails. Ethiopian gender scholarship has primarily centered on child marriage and structural patriarchy, offering limited analysis of how adult women engage with marriage customs. Meanwhile, global feminist literature presents rich critiques of weddings and symbolic gender performances, but these discussions are largely grounded in Western contexts and do not engage in Ethiopia's cultural specificities.

To address these gaps, this study draws on interviews with Ethiopians who identify as feminists and are middle-class, urban, and educated. Their narratives form the basis for a more situated analysis of gendered agency. The next chapter introduces the conceptual framework guiding this analysis, which draws on intersectionality, Nego-feminism, patriarchal bargaining, and Saba Mahmood's theory of embodied agency. Together, these frameworks provide a lens to explore how feminist Ethiopian women negotiate tradition, autonomy, and power within the complex terrain of marriage rituals.

#### 4 Conceptual Framework

This chapter presents the conceptual framework guiding this study, which draws on four key theoretical approaches: Kimberlé Crenshaw's intersectionality, Deniz Kandiyoti's concept of patriarchal bargaining, Obioma Nnaemeka's Nego-feminism, and Saba Mahmood's rethinking of agency.

I selected these frameworks for their critical stance toward the universalizing tendencies of western mainstream feminism. Together, they provide a lens through which to analyze participants' experiences of gendered oppression while also recognizing the culturally and historically specific

#### 4.1 Intersectionality

The first theory I draw on is Kimberlé Crenshaw's (1989) concept of intersectionality, which emphasizes that identities are interconnected, and systems of oppression overlap in ways that cannot be understood in isolation. Developed through her work on domestic violence against women of color and immigrant women in Los Angeles, Crenshaw's notion of structural intersectionality shows how gender-based violence is compounded by race, class, and imigration status. For instance, the U.S. Immigration and Nationality Act required immigrant women to remain with their spouses for two years to qualify for residency, forcing many black women to stay in abusive relationships. Language barriers and economic dependency further exposed them to intersecting forms of racism, sexism, and classism (Crenshaw 1991).

Crenshaw argues that oppression is not additive but interlocking, producing unique forms of marginalization—such as racism experienced by Black women that differs from men of color and sexism that is different from white women. She critiques traditional single-axis approaches for overlooking these complexities and distorting our understanding of both racism and sexism (Crenshaw, 1989)

Building on Crenshaw's insights, I use intersectionality to demonstrate that the participants in my study cannot be understood solely through their gender. Broad and generalized categories such as "woman," "third World," or "feminist" fail to capture the complexity of their identities. Instead, it is necessary to examine their lived realities more closely, with careful attention to other kinds of identities that can reflect the specific social, cultural, and historical contexts that shape how they navigate the world. In this regard, I consider factors such as age, class position, urban residence, and national identity in addition to gender as essential dimensions of analysis. These categories allow me to better understand how intersecting systems of power shape both the constraints my participants face and the forms of agency they express forms of resistance they enact.

#### 4.2 Patriarchal Bargaining

The second theoretical framework I draw on is Deniz Kandiyoti's (1988) concept of patriarchal bargaining, which provides a valuable lens for understanding the interplay between women's oppression and their forms of resistance. Kandiyoti argues that patriarchy is not a universal or monolithic system; rather, it takes on distinct forms across different cultural and historical contexts. Recognizing this variability is crucial, as it allows us to identify the specific "rules of the game"

that shape the strategies—or *bargains*—women make within patriarchal structures (Kandiyoti, 1988).

To illustrate her argument, Kandiyoti identifies two broad models of patriarchy: Sub-Saharan African patriarchy and classic patriarchy, while acknowledging that lived experiences are more nuanced. Classic patriarchy—common in South and East Asia and many Muslim-majority societies—is marked by patrilineal descent, patrilocal residence, and male dominance within the household. In such contexts, women are often married young and transferred from their natal homes to their husband's family, where their subordination is maintained through legal, religious, and familial institutions (Kandiyoti, 1988).

Kandiyoti's framework has been instrumental in analyzing the experiences of marginalized women who are often perceived as lacking agency or not engaging in negotiation. However, her concept of "patriarchal bargaining" reveals that many women do exercise agency in subtle and strategic ways.

For example, research from Kenya shows that some women use silence as a tactic to protect their reproductive autonomy. Akini, a 28-year-old woman, chose not to inform her husband about her decision to use contraception, stating, "It will be my little secret." In doing so, she reclaimed a degree of control over her body and reproductive choices (Gatwiri and Mumbi 2016). Similarly, in Saudi Arabia, women navigate patriarchal marital systems by engaging in contractual negotiations. Some include clauses in their marriage contracts to ensure their husbands will not interfere with their education(Alwedinani, 2017). These examples demonstrate how women work

within patriarchal constraints to assert their autonomy—seeking to influence change from within rather than through overt resistance

I believe Kandiyoti's analytical framework is particularly useful for analyzing the feminist who held the ceremony for two main reasons. Firstly, the Ethiopian context shares many characteristics of classic patriarchy. As outlined in the literature review and background section, Ethiopian society is largely patrilineal and patrilocal, with male-headed households and high rates of early marriage. These structural dynamics are compounded by women's economic marginalization, including limited access to formal employment, land ownership, and financial resources—factors that often make women dependent on male relatives or spouses for financial security. Even the Shimgilina ceremony, the focus of this study, reflects these patriarchal dynamics. It places decision-making power in the hands of male elders—usually the father or senior male relatives—reinforcing gender hierarchies within the marriage process. Secondly, as we have seen in the examples I put earlier, this theory is particularly useful in contexts where women are not typically perceived as having agency, yet in practice, they engage in subtle and strategic forms of resistance or negotiation.

I use patriarchal bargaining as an analytical lens to interpret participants' actions during *Shimgilina* and marriage more broadly. While some choices may appear conformist, this framework encourages a deeper look at what women gain through such decisions.

Given that my participants are urban, educated, middle-class adults, *Shimgilina* may not function as purely oppressive. Their social positioning allows for a degree of agency, prompting questions such as: Why do these feminists choose to participate? What do they stand to gain?

## 4.3 Nego Feminism

The third analytical framework I use in this study is Nego-feminism, a distinctly African feminist approach that offers a culturally grounded lens for understanding gender relations. Coined by Nigerian scholar Obioma Nnaemeka in 2004, Nego-feminism—short for "negotiation feminism"—emphasizes the importance of negotiation, compromise, and community-based strategies over confrontation. It draws from African knowledge systems, lived experiences, and indigenous modes of resistance, making it particularly relevant for analyzing gender dynamics within traditional rituals like *Shimgilina* (Nnaemeka, 2004).

Nego-feminism emerges from a broader pool of African feminist thought, which challenges the universalizing tendencies of Western feminism. Unlike Western models that often frame gender struggles as inherently adversarial, African feminist frameworks—such as Nego-feminism, African womanism, and motherism—emphasize social harmony, relational agency, and the inclusion of men in gender justice efforts. For example, African womanism seeks collective rehabilitation by involving both men and women across diverse social, ethnic, and religious backgrounds (Opara 2024).

This intellectual tradition also critiques the imperialistic and culturally detached approaches of European feminist scholarship. Scholars like Sylvia Tamale argue that Western feminism often "speaks for" African women, reinforcing colonial tropes and denying them agency (Byrne 2020). Other African scholar also notes that African women are frequently portrayed as powerless and dependent on external interventions, rather than as active agents shaping their own transformative agendas (Musingafi 2023).

By anchoring this study in Nego-feminism, I aim to foreground Ethiopian women's culturally embedded strategies of resistance and negotiation, offering a decolonial, context-sensitive analysis of how agency is enacted within marriage rituals.

As previously noted, the foundations of Nnaemeka's nego feminism theory is her critique of dominant feminist frameworks that emerge primarily from Western contexts. She argues that such theories are often disconnected from the lived experiences of African women and lack both cultural relevance and practical applicability. In fact, she labels these approaches as performances of intellectual prestige—what she refers to as "intellectual gymnastics"—rather than as tools for liberation and social transformation. In response, she proposes Nego-feminism: a feminist theory grounded in the logic of African life, focused on the now, and reflected in proverbs, oral histories, storytelling, and community-based ethics (Nnaemeka 2004).

Nnaemeka draws on what she terms the "history of now" to theorize from African ways of knowing and knowledge production, emphasizing that they are rooted in the present. She argues that the present moment can reveal both the origins and the evolution of these knowledge systems. This orientation allows her to build on indigenous foundations and ground feminist theorizing in the lived, everyday experiences of African women—in the now, in what they are doing (Nnaemeka 2004).

Building on this foundation, Nego-feminism not only foregrounds negotiation as a form of agency but also reclaims African culture as a vital and evolving resource for feminist theorizing. Rather than viewing culture as inherently oppressive, Nnaemeka encourages scholars to see indigenous knowledge systems—including cultural practices—as legitimate and generative grounds for theory-building for development. She warns against the wholesale rejection of culture in feminist analysis, emphasizing that although certain practices may be problematic, culture itself is dynamic, continually reshaped by those who live it (Nnaemeka 2004).

I chose this framework because my participants did not completely reject the Shimgilina ceremony, despite its patriarchal dimensions. Instead, they engaged with it in complex, give-and-take ways. Moreover, this approach enables me to focus on their actions in the present—the "history of now"—and to recognize their lived experiences as valuable sites of theorizing and giving them the chance to speak for themselves. Lastly, this framework treats culture not as inherently oppressive—as some strands of Western feminism tend to—but as indigenous ways of knowing and allows for a more complex understanding of culture beyond its patriarchal elements.

## 4.4 Agency Conceptualization

The fourth framework I draw on is Saba Mahmood's (2001) rethinking of agency. In her essay Feminist Theory, Embodiment, and the Docile Agent, Mahmood critiques the dominant feminist assumption that agency is inherently tied to resistance. She challenges the liberal feminist emphasis on autonomy and negative freedom, arguing that agency can also be expressed through self-cultivation, devotion, and adherence to religious norms. Drawing on Foucault's concept of subjectivation—which emphasizes that the same systems that constrain us also shape our identities and capacities—Mahmood contends that if agency is understood as a capacity for action shaped by structures of subordination, it is not solely about fostering resistance or change. Rather, agency

can also be exercised through practices that uphold continuity, stability, and adherence to established traditions (Mahmood 2001).

She supports this argument through ethnographic research with women involved in the mosque movement in Cairo, demonstrating that agency can be expressed through intentional acts of submission and self-discipline. By cultivating practices such as wearing the veil and embodying the virtue of piety, these women do not passively conform but rather actively shape themselves as docile subjects within a religious framework. Through this process, they engage in ethical self-formation—deepening their connection with God and fostering harmony between body, mind, and soul. Mahmood argues that such practices exemplify a form of agency that takes the form of deliberate and conscious docility (Mahmood 2001).

Moreover, Mahmood underscores that many feminist analytical frameworks are deeply rooted in colonial histories and warns against the uncritical application of Western theories to non-Western contexts. To illustrate this, she engages with Judith Butler's theory of performativity, emphasizing that repeated actions are interpreted differently depending on context. For the Egyptian women she studied, the repetition of religious acts—such as veiling and practicing shyness—is a way to align the body, mind, and spirit in the pursuit of virtue. Success is measured by achieving spiritual and ethical coherence through disciplined practice. In contrast, Butler views gender performances as inherently unstable, arguing that failure in repetition reveals the constructed nature of gender itself (Mahmood 2001).

For Mahmood, this theoretical reframing—and the distinction between Western and non-Western contexts—is crucial, as it challenges the tendency to prematurely dismiss or look down upon religious and traditional practices as inherently devoid of agency. By focusing exclusively on resistance, feminist scholarship risks ignoring how individuals actively produce meaning, negotiate values, and engage in ethical self-formation within existing social and cultural frameworks. This perspective is essential for developing more nuanced, non-dogmatic feminist scholarship that is attentive to context and diversity (Mahmood 2001).

I use Mahmood's conceptualization of agency to emphasize her argument that agency is neither universal nor fixed. Rather, it is shaped by the historical, cultural, and social contexts in which it unfolds. Accordingly, the participants in my study must be analyzed within their own specific contexts—including cultural norms, family dynamics, and personal histories—which shape distinct identities, desires and inform how they engage with questions of feminism, autonomy, and negotiation. To deepen this analysis, I build on additional theoretical frameworks that collectively illuminate how agency operates within these layered contexts.

Drawing on intersectionality, I examine how overlapping systems of power—such as gender, class, and culture—interact to shape participants' desires, identities, and capabilities, with particular attention to the influence of tradition and ceremonial practices. Patriarchal bargaining offers a lens through which to explore what participants seek to gain within patriarchal systems, reframing their actions not as passive acceptance but as active, strategic negotiation. Most importantly, Negofeminism positions these negotiations as forms of relational give-and-take, emphasizing that culture is not simply a site of constraint but a space of feminist possibility. It allows for the

balancing of conflicting desires and the development of feminist agency that is embedded in, rather than detached from, local cultural practices.

# 5 Methodology

This study adopted a qualitative research approach to explore the lived experiences of feminists in Addis Ababa who chose to include the Shimgilina ceremony as part of their wedding. The aim is to uncover the nuanced forms of negotiation and agency embedded in their personal stories and decisions. To do this, I conducted semi-structured interviews, a method that encouraged participants to speak openly while allowing me the flexibility to follow up on emerging themes and insights.

As scholars have noted, semi-structured interviews are particularly effective for capturing participants' unique perspectives, as they balance structure with flexibility, allowing the conversation to evolve organically in response to emerging themes (Adeoye-Olatunde and Olenik 2021)

#### **5.1 Data Collection**

Given the relatively small number of self-identifying feminist women in Addis Ababa—particularly those who have had the Shimgilina ceremony—recruiting participants for this study required a multi-pronged approach. Since feminism in Ethiopia is an emerging concept, and is often associated with urban, educated, young, middle-class women, adding a criterion of having experienced Shimgilina further limited the pool.

Nonetheless, I began recruitment with individuals from my personal network, including friends and colleagues who had experienced the Shimgilina ceremony—two of whose ceremonies I had attended. While this provided a few participants, it was not sufficient, so I used snowball sampling and expanded outreach through personal connections and online feminist spaces. This approach was helpful as snowball sampling is particularly effective in contexts where the target population is small and not easily accessible (Naderifar, Goli, and Ghaljaie 2017). Even then, participation remained limited, so I shared calls for participants in a women-only WhatsApp group focused on information sharing and opportunities, as well as in Addis Powerhouse, a young feminist advocacy platform. Some respondents were former collaborators from feminist projects, while others were new connections. In total, I recruited eight participants through this process.

Initially, my criteria focused on feminist-identifying women who had participated in Shimgilina. As recruitment progressed, I broadened the scope to include one woman who adapted the ceremony to her interest, but still who had been deeply involved in a family negotiation, and one feminist man who had chosen to send elders for his wedding, bringing the total to ten.

The semi-structured interview followed a two-part section. The first part focused on collecting personal background information, including age, gender, religion, education, nationality, profession, and place of residence, as well as how participants identified as feminists. This information allowed me to analyze their experiences through an intersectional lens and to contextualize their stories, negotiations, and expressions of agency within their broader social identities.

The second part of the interview consisted of open-ended questions accompanied by prompts. As scholars note, a semi-structured interview guide typically includes core open-ended questions along with follow-up probes to help the interviewer explore emerging themes throughout the conversation (Adeoye-Olatunde and Olenik 2021). Thus, I asked participants how they defined the *Shimgilina* ceremony, what their decision-making process entailed, how their ceremony unfolded, whether they attempted to change any aspects of it, and their reasons for doing so or not, with probing questions when needed

The Interviews were held in flexible, participant-selected settings to ensure comfort and privacy. These included homes, cafés, and co-working spaces. In some cases, interviews were conducted in domestic environments where participants juggled conversation with me along with caregiving responsibilities with their kids, while others took place more casually over coffee. Each interview lasted between 40 and 60 minutes. I took notes on my notebook and recorded conversations using my phone. The interviews were conducted in a mix of English and Amharic. I transcribed and translated all recordings into English.

## 5.2 Data Analysis

For data analysis, I used thematic analysis, starting with repeated readings of transcripts to identify key phrases and patterns. This process revealed two participant groups: those who reinterpreted Shimgilina and those who aimed to reform it. This method proved valuable for capturing the complexity of participants' experiences, as it facilitated a comparative lens through which both shared narratives and contrasting viewpoints could be analyzed (Nowell et al. 2017).

#### **5.3** Ethical Considerations

In qualitative research, informed consent is understood as an ongoing process of building trust between the researcher and participants. It involves providing enough information without overwhelming participants, while respecting their right to freely choose whether to take part in the study (Orb, Eisenhauer, and Wynaden 2001). To uphold this, I shared a written consent form—detailing the study's purpose, methods, duration, and confidentiality measures—with most participants before each interview, along with the guiding questions. At the beginning of each interview, I reiterated these points verbally, assuring participants of their right to skip questions or withdraw and that pseudonyms would be used to protect their identities.

Due to the sensitive political and cultural climate in Ethiopia, I intentionally avoided questions related to ethnicity and chose choose not to explore topics related to sexuality, for safety and ethical reasons.

## 5.4 Positionality

To reflect on my positionality as a researcher, I draw on feminist scholarship that highlights how our social location shapes research, relationships with participants, and knowledge production. As emphasized by *Affilia: Feminist Inquiry in Social Work*, acknowledging positionality is key to ensuring transparency and reflexivity throughout the research process (Jackson et al. 2024).

Guided by this approach, I reflect on key feminist questions about my relationship to the research topic and participants, the power dynamics shaped by my social position, and how these should be

represented. These reflections have shaped my reflexive practice and informed my role in the study (Jackson et al. 2024).

I am Kalkidan Asmamaw, a 28-year-old Ethiopian woman, born and raised in Addis Ababa. I speak Amharic, identify as a feminist, and am currently pursuing a master's degree in Gender Studies at Central European University. Although I am unmarried and have not personally undergone the Shimgilina ceremony, I am closely familiar with the cultural context in which it is practiced.

Over the past years, I have attended multiple Shimgilina ceremonies—dressing in coordinated outfits, celebrating with friends, and sharing congratulations on social media—I am part of the world where this tradition holds meaning. I also share key social characteristics with most of my participants: we are urban, educated, feminist Ethiopian women(of a similar age. In qualitative research, such shared identities often define an "insider" positionality, enabling ease, trust, and openness in the interview setting (Bukamal 2022).

However, insider status is never fixed. As scholars note, "all cultures (including subcultures) are characterized by internal variation," and insider/outsider lines are often blurred remained (Merriam et al. 2001). While I shared much with my participants, key differences still remain.

First, although we all lived in Addis Ababa, we came from diverse ethnic backgrounds. As discussed in the literature review, ethnicity in Ethiopia often carries distinct cultural meanings and practices, including variations in how marriage rituals like Shimgilina are understood and

performed. These differences, while sometimes subtle, shaped both how participants narrated their experiences and how I was perceived.

Second, my affiliation with a European academic institution introduced a layer of epistemic authority. My role as a researcher—particularly one trained in feminist theory abroad—sometimes positioned me as a bearer of external or "Western" knowledge. This complicated the horizontal nature of the research relationship. In some cases, my background as a gender studies student seemed to shift the dynamic; I was seen as an "expert," and a few participants appeared to seek validation from me. These moments made me acutely aware of the power I held in shaping the conversation. In response, I tried to make clear that this was their story, and they were the experts of their own lives and choices.

Conversely, with a few participants, especially those who were older or more professionally established—I sometimes felt unsure of myself or hesitant to assert my role. In these instances, my identity as a younger researcher positioned me differently. This was further reinforced by differences in life stage: some participants were already married, had children, or were managing family life, while I remain unmarried. I occasionally sensed that participants were adjusting their narratives—either to include me or to explain decisions they assumed I might not fully understand.

These shifting dynamics revealed that my positionality was not static. At times I was perceived as a peer, as an outsider, a student, or a presumed expert. I navigated these roles throughout the research process, recognizing that insider and outsider positions are always relational and fluid.

#### 5.5 Limitations

One key limitation of this study is the specificity of its participant group. All interviewees were urban, highly educated, and middle-class individuals based in Addis Ababa. As a result, the findings reflect the perspectives of a relatively niche demographic—feminist-identifying individuals with access to certain social and economic resources. This focus, while intentional, limits the generalizability of the study to the broader Ethiopian context. In particular, the experiences and negotiations of women from rural areas, lower-income backgrounds, or with limited access to formal education may differ significantly from those captured here.

## 5.6 Methodological tools

## **5.6.1 Feminist Methodology**

I position my work within feminist methodology not simply because I interviewed women, but because of the decisions I made regarding method, methodology, and epistemology. In terms of methods, I drew on conventional qualitative tools, such as semi-structured interviews, listening, and observation. However, following feminist methodology, what makes this study feminist is my effort to identify emerging patterns of recognition and meaning-making in feminists narratives (Harding, 1987). Existing research on the *Shimgilina* ceremony rarely analyzes it through a gendered lens in the way that I do. It is typically framed in a polarized understating either in discourses of child marriage or understood as a celebrated cultural tradition. My approach, by contrast, centers on the gendered performances, acts of resistance, and expressions of agency demonstrated by feminist women who *consented* to participate in the ceremony. Through this lens,

I aim to uncover the nuanced complexities of their agency in cultural settings moving beyond simplistic interpretations of acceptance or resistance.

#### 5.6.2 Intersectionality as a Method

My central argument is that self-identified feminists choose to include Shimgilina as part of their wedding, and this choice should be understood as a complex, context-specific expression of agency. However, their actions cannot be fully understood without considering the context and their intersecting identities, which inform both the nature of the oppression they face and the forms of resistance they employ.

As Kimberlé Crenshaw (1989) argues, gender discrimination cannot be examined in isolation from other social categories such as race and class. These systems of power intersect to produce distinct and compounded forms of marginalization. Crenshaw emphasizes that relying on a single-axis framework obscures the complexity of oppression, resulting in what she calls "a distorted analysis of sexism" (Crenshaw, 1989, 140). Therefore, to fully understand how sexism operates and manifests within a specific context, it is essential to incorporate these intersecting dimensions into our analytical frameworks. Moreover, this way of analyzng identities is crucial in exploring how oppression is challenged, or when it is being accepted.

To lay the groundwork for my argument, I will first use intersectionality, not just as a theoretical framework but as a method as well. I applied intersectionality as a research method by intentionally asking about categories beyond gender, starting with the questionnaire. Following Helma Lutz's (2015) call to "ask the other question"-that is, to always consider which additional axes of power

are at play (Helma Lutz 2015), I tried to move beyond merely identifying sexism to also examining how class, age, and other intersecting factors shape my participants' experiences. When something appeared sexist, I asked: *Where is class in this? Where is the race in this?* My goal was not to treat these categories in isolation or mention them superficially, but to explore how they interact and co-construct the realities my participants navigate.

Another way I incorporate intersectionality as a method is by paying close attention to how participants themselves highlight specific aspects of their identity while narrating their experiences. Do they emphasize their gender, age, ethnicity, or national identity? When do these identifications surface, and in what context? This approach is also inspired by Helma Lutz, who reminds us that identity is not fixed or uniform—it shifts across different contexts and situations, and must therefore be examined as fluid, relational, and shaped by varying social dynamics (Helma Lutz 2015).

As noted in the earlier section of the methodology, I focus on five key dimensions: age, gender, class, urban residence, and national identity. Before exploring how these categories intersect, it is important to first understand what each of these identities represent individually, particularly within the context of this study and especially regarding in marriage ceremonies. This allows for a clearer analysis of how they interact and shape the negotiation strategies and expressions of agency among my participants.

#### 5.6.2.1 Gender

Nine of the ten participants in this study are women, with one male participant as the exception. In the Ethiopian context, being a woman often involves navigating structural disadvantage within a socio-political landscape shaped by persistent gender inequality and deeply entrenched patriarchal norms. On a global scale, Ethiopia ranks 173 out of 188 countries on the Gender Inequality Index, highlighting critical deficits in reproductive health, empowerment, and economic participation (Wright, 2020.). Local data reinforces this picture: according to the 2016 Ethiopian Demographic and Health Survey, 63% of women and 28% of men believe that wife-beating is justified under certain circumstances (Kelecha et al. 2024). In the interviews, all participants spoke about the oppressive systems they face highlighting the women part of them is often oppressed

Tigist, a 28-year-old radical feminist who specializes in gender and media, said that

"Ethiopia is very traditional when it comes to gender roles. You can't be radical here—I can't exist in this community. You can't get married, make friends, or have a relationship. You can't even get a job." (Tigist, 28)

Participants also expressed forms of resistance in response to these oppressive structures, especially by trying to influence others around them. For instance, Soliyana, who is a female researcher and development worker, told me that

"I always challenge the patriarchal assumptions of my family—even my mom and siblings. Sometimes it's just when we're watching TV and an ad comes on, and I point out what's wrong with it." (Soliyana, 34)

#### 5.6.2.2 Nationality

Secondly, all these participants are Ethiopian —a designation that, I argue, carries significant meaning. As discussed in the background chapter, Ethiopia's legacy of resisting colonization and its emphasis on cultural continuity have shaped a powerful national discourse rooted in pride, tradition, and self-reliance (Hailu 2021; King and Schramme 2019). Within this context, culture transcends mere practice; it symbolizes sovereignty deeply embedded in anti-colonial narratives. Consequently, marriage ceremonies such as *Shimgilina* serve not only as familial rituals but also as affirmations of social continuity and cultural resilience (Dessie & Bekelcha, 2023)

This sense of cultural continuity is reflected in the participants' narratives, particularly when they explained why they chose to engage in the Shimgilina ceremony. Their accounts reveal that the ritual holds a deeper significance beyond the act itself—it is embedded in their understanding of identity and belonging. One participant stated, "I initiated it because my mother and aunts went through it," positioning her decision within a matrilineal tradition. Others referred to Shimgilina as "our culture" or described it as "the way you introduce him to the family," signaling a collective identity rooted in Ethiopian or ethnic heritage. These expressions suggest a strong sense of communal affiliation—a "we" that participants feel part of and wish to preserve. Thus, for some of the participants, their engagement with the ceremony, therefore, is not just personal but also a meaningful affirmation of cultural continuity and national identity.

That said, it is important to acknowledge that Ethiopia's anti-colonial legacy, while preserving local traditions from direct colonial interference, has not led to a unified national culture. Instead, Ethiopia is home to over 80 ethnic groups, each with its own language, customs, and cultural

practices. This diversity—compounded by ongoing ethnic tensions and political instability—has made ethnic identity a dominant framework of social belonging (Birhan, 2024). Given this context, it is likely that references to "we" in participant narratives may also reflect ethnic belonging rather than national identity. However, because I intentionally avoided asking about ethnicity due to its sensitivity, I have chosen to interpret "we" in terms of national identity for this analysis. Ultimately, it is important to recognize that cultural practices such as *Shimgilina* are not merely traditional rituals but also serve as potent expressions of cultural sovereignty and/or ethnic identity.

This conceptualization of culture offers a powerful means of self-expression, and connection with the older generation. However, it is also a space where gendered power relations and social hierarchies are reproduced. While many participants expressed a strong sense of cultural belonging and appreciation for tradition, they simultaneously sought to challenge or transform certain aspects of the ceremony. Their efforts, however, were often constrained by the intersecting pressures of gender and culture/ethnic identity—factors that frequently work in tandem to suppress or delegitimize young feminist voices within these cultural contexts.

Tsige, a 28-year-old consultant working in the gender sector, was determined to include women in the ceremony by having them sit on her side of the family. However, her request was denied by her parents. In response, she turned to her fiancé's family and asked if they could at least include a woman among the elders." As she recalled firmly: "I asked my husband's older sister if they could include a woman among the elders, and she flat-out said, '[X ethnic] men don't do that.' After that, I just staved quiet."

This example shows that Tsige was actively seeking a space to assert her voice. However, her attempt was unsuccessful, as she was challenging a cultural practice deeply tied not only to expressions of ethnic identity but also to the ways in which certain ethnicities construct and perform masculinity. Furthermore, the fact that she was a young woman—and that her husband's older sister held more authority due to age—further undermined her efforts. Confronted with these significant obstacles, she ultimately felt compelled to remain silent.

#### 5.6.2.3 Capital City Residence

All the participants reside in Addis Ababa, which I believe is closely linked to their high level of educational attainment. This urban residence reflects a certain degree of social advantage. Historically, in the Ethiopian context, as more young people gained access to education and migrated from rural areas to urban centers, their autonomy and life choices expanded. For young women, moving to urban cities and education enhanced their decision-making power in terms of marriage. Moreover, Western influences at urban areas introduced new social spaces—such as school plays and sports activities—where young people could meet and form relationships in their own terms. These changes not only facilitated the incorporation of Western elements into cultural practices but also created opportunities for young women to exercise greater agency (Mengesha, Deressa, and Imagnu 1996).

This broader context helps explain why participants in this study viewed their own Shimgilina ceremonies as evolving cultural expressions, not solely due to their feminist perspectives, but also because of their exposure to their urban residence. They often drew the line between urban/rural, traditional/modern, and past/present

Soliyana, who is based in Addis Ababa, reflected on the evolution of the ceremony:

"Historically, the ceremony involved family negotiations without the woman's knowledge or consent. But today, it's more of a symbolic step for acknowledgment and blessing, rather than a strict requirement." (Soliyana, 34)

Tigist, who was born and raised in Addis Ababa, emphasized the shift in agency between the modern and traditional: "In the modern sense, the woman chooses who she wants to marry."

Moreover, Tsige, who had her Shimgilina after reaching an agreement with her husband, noted:

"Traditionally, the woman is not involved in the process, but in the urban context, both the man and the woman agree beforehand." (Tisge, 28)

Participants often distinguish their own *Shimgilina* ceremonies from past, traditional, or rural versions by framing theirs as modern, urban, and the present. While they acknowledge that in other contexts *Shimgilina* may occur without the woman's consent, they emphasize that their participation involved mutual agreement and personal choice. This reinterpretation reflects not only their awareness of structural inequalities but also their active role in reshaping cultural practices to align with their values and lived realities.

#### 5.6.2.4 Class

Another key aspect of the participants' identities that affords them a significant advantage is their middle-class status. All of them identify themselves as in the middle class and nearly all of them hold master's degrees or higher and work as professionals in roles such as consultants, researchers, and program managers within both local and international NGOs. This positions them firmly within Ethiopia's middle class—a considerable level of privilege when compared to the broader

population of women in the country. As discussed in the literature review, most Ethiopian women have limited access to education and live in conditions of poverty.

#### 5.6.2.5 Age

Age is a crucial factor in assessing the bargaining power of my participants. In shaping the Shimgilina ceremony, they were often negotiating with parents and "elders" who were significantly older than they were. This generational gap inevitably constrained their ability to assert influence. Especially in Ethiopian society, older individuals are generally seen as bearers of wisdom, experience, and authority, especially when it comes to the preservation of cultural traditions. They are typically entrusted with upholding such practices, which further reinforce their dominant role in decision-making (Kassas, and Tessema, 2019). In contrast, the participants in my study were relatively young at the time of their ceremonies, most of them in their twenties, with the oldest being 31. This age disparity played a meaningful role in shaping the dynamics of negotiation and agency.

## 5.6.3 Feminism in Context: Intersecting Realities of Ethiopian Feminists

Due to their identities that are context-shaped and the nature of the intersection. I believe it is crucial to approach this group with careful consideration. For instance, we cannot analyze them solely through frameworks that prioritize gender as the primary axis of oppression, as radical feminism often does (Rowland and Klein 1996), as such an approach overlooks the complexity of their lived realities. In the Ethiopian context, gender intersects in distinct ways with poverty, religion, postcolonial identity, and ethnicity—dimensions that are not fully addressed by theories rooted in countries like the United States, Germany, or France (Rowland and Klein 1996). As a

result, analyzing gender oppression and resistance without accounting for these intersecting factors risks flattening the nuanced ways in which Ethiopian women navigate both their oppression and their cultural traditions.

To offer a more specific example of the contrast between radical feminism and the interests of Ethiopian feminists, radical feminism calls for a total transformation of social structures and often views the elimination of patriarchal practices as essential(Rowland and Klein 1996). However, for women in Ethiopia, the reality is more complex. Although there is no formally articulated framework of Ethiopian feminism, the data from this study show that none of the participants expressed a desire to eliminate the *Shimgilina* ceremony.

Mare, who did not assign a specific label to her feminism but felt most aligned with African feminist perspectives, explained:

"It's about who we are, and we should continue it. If it doesn't have a real negative implication, it's actually a positive thing—you know, it's about making the elders happy, making your parents proud, and taking pride in the ceremony. But we should completely abolish it if the couple can't give consent." (Mare, 38)

Soliyana, who identifies as a radical feminist, similarly articulated a vision of transformation rather than elimination: "Even in the future, if I have kids, of course the father will be involved—but so will the women in her life.

Whether they identify with African or radical feminist perspectives, these women still express a desire to preserve the ceremony. As the interview excerpts suggest, they do not seek to discontinue it. While they offer critiques, such as the need for consent and greater inclusion of women—they

do not advocate for its abolition. For many, the ceremony represents a form of cultural identity, or simply something so deeply embedded in their lives that they cannot imagine removing it.

This example helps a potential disconnect between dominant feminist theories and the lived experiences of feminists in Ethiopia. Most participants did not identify with established academic categories such as radical, liberal, or African feminism. Instead, they framed their views through personal experience and everyday struggle. When asked, the only widely recognized feminist figure they mentioned was Chimamanda Ngozi Adichie.

Eden who is a 29 year old told me that:

"Growing up, I used to get punished because I didn't obey when they ordered me around, saying, 'You're a girl.' I used to fight with my brothers—I didn't like to lose anyway. Then, when I got to university, I realized how bad it is for women, and I started calling myself a feminist." (Eden, 29)

Fatima, who learned feminism from the strong women in her life:

"I didn't really know there was such a big problem when it comes to gender equality because the women who raised me didn't tell me as such. It was only when I joined university that I realized that there is a problem with being a woman." (Fatima, 30)

The feminists in my study did not arrive at their feminism through alignment with major theoretical frameworks such as radical feminism, poststructuralist feminism, or even nego-feminism—many of which I was not familiar with before beginning this academic journey.

Instead, they theorize from their everyday lives—through their interactions with others, their personal struggles, and the realities they navigate daily. The For example, Eden's feminist awakening was closely tied to her exposure to new discourses during her university years, while Fatima traced her feminist consciousness to the strength and wisdom of her mother and

grandmother. These narratives exemplify the practice of theorizing from lived experience, where personal and intergenerational insights inform broader critiques of gendered power structures.

Accordingly, this research intentionally grounds its analytical approach in the lived experiences and meaning-making practices of these Ethiopian feminists as a continuation of their feminist journeys. Rather than imposing dominant feminist frameworks developed in different sociopolitical contexts, I draw on theories that create space for and incorporate the ways in which participants themselves narrate, negotiate, and embody their feminist commitments. In doing so, this study contributes to a contextually grounded articulation of Ethiopian feminism—one that emerges not from abstraction, but from everyday struggles, strategies, and acts of resistance, thereby contributing to the participants' own conceptualizations of agency.

# 6 Navigating Tradition:

# The Emergence of Two Feminist Approaches to Shimgilina

This chapter presents the core analysis of my research, examining how self-identified Ethiopian feminists in Addis Ababa engage with the Shimgilina ceremony—a traditional, patriarchal prewedding practice. Drawing on the conceptual frameworks introduced earlier, I explore how participants negotiate the tension between feminist commitments and cultural expectations. Their narratives reveal that agency is not expressed solely through resistance, but also through adaptation, negotiation, and cultural embeddedness.

While the participants share common social positions—such as age, gender, nationality, class, and urban residence—their approaches to Shimgilina vary significantly. For some, agency was rooted in the freedom to choose their partners, reducing the perceived need to challenge the ceremony itself. For others, Shimgilina became a site of active negotiation, prompting efforts to reshape its structure and meaning.

I argue that both approaches represent valid forms of agency. Each participant navigates his/her struggle in ways that reflect her lived experience and the tools available to him/hers

## 6.1 Reinterprets: Making Feminist Sense of Tradition

"As long as I made sure they couldn't say no to me about who I am going to marry, it was fine. It's what tradition expects, and from what I saw when my aunt got married, it was a source of great pride for the family, so I wanted to honor that." (Eden, 29)

Rather than challenging the structure of the ceremony, participants from this group chose to engage with it in its existing form. The following analysis explores the reasons behind this approach and the actions that accompanied it. I argue that their decision not to change the ceremony should be understood as a strategic and agentic choice, shaped by their sense of autonomy, the cultural value they attribute to the ceremony, their privileges, and the dynamics within their families.

## **6.1.1** Feminist Critique by Reinterpreters

"It's usually men who sit and ask the questions. Even in my case, it was my uncle and my dad who did that." (Eyerus, 38)

The participants in this group, many of whom explicitly identified as feminists, were quick to recognize the patriarchal dimensions of the Shimgilina institution. Their awareness extended beyond gendered power dynamics, allowing them to critically examine other intersecting systems of oppression, such as ageism and classism.

This depth of analysis can be attributed, in part, to their backgrounds as highly educated professionals in Addis Ababa, many of whom work in the fields of gender, human rights, and social justice.

Petros, a 41-year-old man currently pursuing a PhD at a prestigious international university, had his own *Shimgilina* approximately a decade ago. During the interview, his engagement with the

ceremony felt emotionally distant; rather than reflecting on his personal experience, he spoke more about how things are, offering a detached, systemic, and analytical account of the ceremony. I believe he was drawing more on his academic training than on his lived experience during our interview.

Nonetheless, his reflections offered a critique of both the gendered and class-based dimensions of the ceremony. He emphasized that Shimgilina is not only patriarchal but also deeply shaped by class hierarchies, particularly through its focus on the financial capability of the groom. I believe his positionality as a man enabled him to critique both the pressure placed on men within this system and how men may perceive their wives because of this tradition.

He acknowledged: "Even men from disadvantaged and underprivileged backgrounds might struggle to marry or be with a woman they love because of traditions like this." At the same time, he was critical of how these practices harm women as well, stating: "This arrangement is also bad for women—he might end up seeing her as furniture that he paid for."

Petros had his Shimgilina at a time when urbanization and globalization had not yet significantly influenced the ceremony. Reflecting on more recent developments, he emphasized that Shimgilina is not a static tradition but one that is gradually transforming. He cited a recent example of a ceremony organized entirely by women and widely circulated on social media—an event he described as "shocking." At that moment, he explained, it prompted him to reflect on the gendered language embedded within the tradition critically. He pointed to the asymmetry between the term shimagile—a title reserved for elder men and associated with wisdom and respect—and arogete, the supposed female equivalent, which he noted is often used in a dismissive or derogatory manner.

Eden, a 29-year-old university graduate who lives in the capital city with her husband and is currently working multiple jobs, spoke with firmness when I asked whether she had any objections to *Shimgilina*. Rather than relying on formal education—she mentioned she had no academic background in feminism—Eden drew on her empathy and everyday feminist sensibilities, noting that she learned much of her feminism from Ethiopian TikTok. In her response, she was visibly empathetic toward the girl she was describing as being affected by the ceremony. She focused sharply on the woman's experience within the tradition, portraying it as a process in which the bride is rendered passive and victimized. "*She's not consulted*," Eden said. "*She's targeted. Others have already decided for her—and the next thing she knows, she's a wife.*"

In Eden's case, she initially viewed her experience as more modern. Laughing, she remarked that she wouldn't even let her parents choose something minor on her behalf. With a smile, she added, "So why would I let them choose something this big?" However, as the conversation shifted toward a critique of *Shimgilina*, her tone became more serious—she grew increasingly reflective and critical.

Similarly, Abigael, a 33-year-old who holds a master's degree in human rights from a foreign university, spoke about *Shimgilina* in a very structured and analytical way. She focused on how the ceremony gives power and visibility to men. "*Elder men are the ones who ask the questions*," she said, also pointing out that "*it's always the man who sends elders to propose*." For her, these actions are not just traditions, but ways that men show and confirm their authority. She explained that if a woman were to start the ceremony herself, it would be seen as disrespectful and would make the man look weak. "*No one would agree to go*," she said. Throughout the interview, Abigael used terms like "*agency* and *masculinity*" to describe how the ceremony is built to support male power and keep women in passive roles.

These reflections reveal that the ceremony is not accepted uncritically. Participants recognized that *Shimgilina* operates within a patriarchal system in various ways—one that legitimizes male authority, objectifies women, and limits their agency and representation.

For Petros, his male positionality allowed him to focus on the pressure placed on men to demonstrate economic capability. And this is accurate, as research shows that bride wealth is often part of the *Shimgilina* ceremony (Balew Liyew 2024). Eden, on the other hand, centered the experience of the woman who lacks agency in the process. She refers to cases in which parents decide without the girl's consent (John et al. 2019), often to strengthen their families (Abera et al. 2020), or to protect her from premarital pregnancy or gender-based violence (Abdurahman, Assefa, and Berhane 2022) For Abigael, the core issue was the structural power granted to men within the ceremony, which is in line with global literature that critiques the authority given to fathers or male figures—such as giving the bride away (Froschauer and Durrheim 2019), or the groom asking the father for permission (Fairchild 2014).

Nonetheless, it is important to note that participants did not problematize the gendered labor of their female relatives, who were typically responsible for preparing food and managing logistics. All of them mentioned that it was the bride's female family members who served food to the elders after the ceremony, yet none identified this as an issue. This recurring pattern reflects a broader theme observed in global feminist conversations, but not in this research in the Ethiopian context (Schweingruber, Anahita, and Berns, 2004).

## 6.1.2 Social Bonds and Cultural Legitimacy: Rewriting the Rules of Shimgilina

"It was important to them, and I wanted to respect that. That is why I had the ceremony. By that time, my father was no longer around, and I wanted my mother to have the same respect, even without him. I didn't want her authority in the family to be negated." (Abigael, 33)

Abigael, for example, while critiquing the ceremony for perpetuating gendered power structures, also acknowledged its potential to foster social bonds, grounding her argument in an indigenous understanding of marriage and authority. As she explained: "When two people get married, their families also come together to make future decisions—like about the wedding." She also shared that while her sister had received Shimgilina when their father was still alive, her ceremony took place after his passing. In this context, she used the ceremony to affirm her mother's authority within the family.

"I participated in Shimgilina to build a good relationship with my in-laws and because my partner wanted it." (Petros, 41)

For Abigael, the decision was easy. However, for Petros, it wasn't a straightforward choice. He mentioned that all his friends had done it and that no one he knew had ever skipped the ceremony. This made him feel pressured to go along with it. As he said, "I have been conscious about these things for the past, I don't know 10-15 years. I never came across my circle. I have never met someone who did not do it." There was a quiet anxiety underneath—he didn't want to be the one who broke from what was expected. For Petros, taking part wasn't about agreeing with the tradition, but about keeping peace with his partner, building a good relationship with his in-laws, and fitting in with what everyone else was doing.

Eden, who described *Shimgilina* as a form of arranged marriage, still recognized its emotional and symbolic value. "When a man respects you, he also respects the people who shaped you," she said, explaining how the act of sending elders can be seen as a gesture of love and respect, not just toward the bride, but toward her entire family. I believe this interpretation was rooted in her experience observing her cousin's ceremony from beginning to end. She recalled how, during the

event, the groom's elders were made to wait outside before being invited in and receiving an answerable moment she described as symbolically powerful. "It highlights the value of your womanhood," she reflected. For Eden, these gestures, though deeply embedded in patriarchal structures, were also understood to be affirmations of a woman's worth and sources of family pride.

The quotes reveal that *Shimgilina* functioned not only as a patriarchal ritual but also as a meaningful practice that promoted social cohesion. For these participants, the ceremony served to strengthen relationships—with partners, in-laws, and extended family—underscoring its role as both a cultural and relational institution. I believe this perspective aligns with viewing the ceremony as a means of facilitating bonds between families, as discussed in the ethnographic accounts of the ceremony in the literature review (Worku Kalore 2019).

For others, the ceremony was viewed as the official and culturally appropriate way to formalize a relationship, particularly within families and communities where such rituals carry deep social significance. These sentiments were expressed with pride, not reluctance or regret; participants stood by their decisions with a sense of cultural

Mare, who critiqued the ceremony for excluding women and called for greater female participation in the decision-making process, also reflected on its cultural significance. "This is our identity, and even the idea of a proposal is a Western concept—this is our way of doing things," she said. While she now views Shimgilina as an important expression of cultural identity, she admitted that at the time of her ceremony—about 15 years ago, when she was 24—she didn't give it much thought, as she was young. She was already living with her now husband, and although her mother was aware of the relationship, her father, as she put it, "pretended that he did not know." This was because the relationship hadn't yet been made official. In her words, Shimgilina was how the

family formally acknowledged the union—only after the ceremony did her father say, "*Now you* can live together." For Mare, the ceremony was simply a way to legitimize the relationship in the eyes of her family, especially her father.

Similarly, Eyerus—a 38-year-old mother who holds a master's degree and currently lives in Addis Ababa, though she was born and raised in another town—described *Shimgilina* as "the official way of showing love and letting others know." At the time of her own ceremony, her siblings knew about her boyfriend, but her parents did not. For her, *Shimgilina* was the formal and culturally appropriate way to introduce him to her parents. She noted that while she might not have a wedding or other ceremonies, this was the meaningful way she chose to present her relationship. She claimed that "this was the way that I was raised."

Participants perceived the *Shimgilina* ceremony as a meaningful expression of their identity, referring to their Ethiopian identity, something they took ownership of and spoke about with pride. Moreover, by upholding the tradition in this way, they were able to gain social recognition and legitimize their relationships with their partner among their communities. This perspective—viewing marriage ceremonies as expressions of identity—is also supported by ethnographic research cited in the literature review (Beyene, 2006).

## 6.1.3 From Oppression to Expression: Fighting Patriarchy with Autonomy

"It was just for show. Whether they approved or not, I knew I was going to marry him. It was just a performance, but for my family, it held real meaning." (Mare, 38)

The participants openly criticized the systems of oppression embedded in the ceremony. Nonetheless, several patriarchal elements they identified as problematic persisted in their celebrations because of the culture. For instance, all of them noted that "he was the one who sent

the elders," reinforcing the traditional gender norm that positions men as the initiators of marriage. This is in line with other similar research that documents the Shimgilna ceremony across Ethiopia, with men taking the lead in the process. Either the groom identifies a girl to marry and informs his father (Telila, 2020) or the parents of the groom identify a "suitable bride" for their son (Beyene, 2006) Additionally, in each case of my participants, the male partner was represented by older male relatives who spoke and acted on his behalf. The elders from here side asked, "Can he provide for her?"—a question that underscores the enduring patriarchal assumption that financial responsibility rests with the man.

Even now, some participants are currently actively reproducing the tradition. Petros, for example, laughed about attending a *Shimgilina* the following week as an "elder." While lighthearted, the moment revealed how age and masculinity grant authority within the ritual, positioning him as both a critic and a carrier of the very structure he critiques.

As outlined in the literature review, Shimgilina in rural contexts is often characterized by limited female agency, with parental authority dominating decisions (Gebeyehu et al. 2023) and the exchange of bridewealth playing a central role (Balew Liyew 2024). In some accounts, the ceremony is equated with arranged marriage and is closely linked to child marriage, frequently involving girls under the age of 18 (Abera et al., 2020). In sharp contrast, none of these elements were present in the participants' experiences. All were over the age of 24 at the time of their ceremonies, had full autonomy in choosing their partners, and did not engage in bridewealth exchanges. These contrasts underscore how Shimgilina is being selectively adapted—preserving certain cultural forms while discarding others. This process reflects the participants' complex negotiations with culture, patriarchy, and modernity, and illustrates how feminist agency can be expressed through careful and strategic reworking of tradition.

For this group, the absence of traditional constraints, particularly those common in rural settings, such as parental control over partner selection and the timing of marriage, was especially meaningful. The ability to make these decisions independently significantly enhanced their sense of personal autonomy.

Mare, for example, had already been living with her now-husband before the elders were sent. She met and began a relationship with him while studying in another city, without any involvement from her parents. Reflecting on her agency, she shared: "It was just for show. Whether they approved or not, I knew I was going to marry him"

To further support the point that she independently chose her husband, she shared, after the interview, that her family had initially been deeply anxious because her partner belonged to an ethnic group they were unfamiliar with. "They were really scared about what he might even look like," she admitted.

For another participant, Eden, there was no parental control over her marital choices. Reflecting on her decision to proceed independently, she stated: "If I made sure they couldn't say no to me about who I'm going to marry, it's fine." She explained that she had already moved out of her parents' house, lived alone for two years, and had been living with her partner for eight months—without her parents knowing.

From my observations, it appears that this group exercised a significant degree of autonomy well before the *Shimgilina* ceremony took place. They chose their own partners and engaged in relationships where they got to know one another as boyfriend and girlfriend—without parental involvement. In fact, three out of the five participants reported that they had already been living with their partners in some cases in a different city prior to the ceremony, without their parents'

knowledge. Ultimately, they made the decision to marry their partners on their own terms, with or without a formal proposal from the man (which is a separate topic deserving its own discussion). In some cases, they even cross ethnic boundaries in choosing a partner. Taken together, these experiences reflect a strong sense of personal autonomy, with participants placing considerable value on their ability to make independent life decisions.

Eyerus, who chose to get married at the age of 31, was the one who initiated the *Shimgilina* ceremony with her now-husband. She first informed her sister, asking her to pass the message on to their parents. After receiving the news, her parents worked with the couple to choose a date that fits the couple's work schedules. Once the date was set, her parents informed her of the time the groom's family should arrive—6 a.m. in the morning. When they came, Eyerus's father, mother, and other relatives were present to receive them. They asked the customary questions, held the initial conversation, and then told the groom's family to return later, following tradition.

"My boyfriend and I decided we wanted to make our relationship official, and then my parents took over—deciding who they would call, when the date would be, and handling all the ceremonial details." (Eyerus, 38)

This suggests that the decision to marry was primarily their own, made independently of parental pressure. In each case, participants first decided with their partners to hold a *Shimgilina* ceremony, and only then informed their parents. While they had full control in choosing their partners, their parents retained decision-making power over the ceremony itself—determining who the matchmakers would be, who would sit on each side, what questions would be asked, who to invite, the date of the ceremony, and even the food that would be served.

For this group, what mattered most was the partner they had chosen, the timing of formalizing the relationship, and the future they envisioned together with their partner As a result, many perceived

the remaining gendered aspects of the Shimgilina ceremony as having minimal impact on their lives. While they acknowledged its patriarchal dimensions, they often described the ceremony as "shallow," "theatrical," "dramatic," or "merely performative," indicating that, for them, its significance was more symbolic than truly consequential.

I believe this group perceived the ceremony as largely symbolic, something that did not exert real power over them. They contrasted their experience with more traditional rural practices, where it is common for a girl to be entirely under her parents' control and to have little or no voice in the decision-making process (Erulkar 2013; Gebeyehu et al. 2023)

Petros, who had sent male elders for the Shimgilina ceremony and had been married for eleven years, remarked: "Even though we had a ceremony like Shimgilina, I still change diapers, we clean the house together, and we try to divide responsibilities equally." His reflection highlights how the symbolic performance of tradition did not dictate the dynamics of their everyday marital relationship.

Even in the case of Eyerus, who now lives in Addis with her husband, the Shimgilina ceremony was held in her hometown to accommodate familial expectations. Because her husband had no connections in that area, it was her parents who selected the elders to represent him. "They ended up choosing my father's friends," she explained, "just so we could say someone was sent."

In her experience, it wasn't even real male elders who participated—it was a family friend. This made the ceremony feel somewhat superficial to her and reinforced the sense that it did not hold real power over her.

Based on the information above, one can argue that although the participants were fully aware of the patriarchal aspects of the *Shimgilina* ceremony, other factors influenced their decision to participate without necessarily attempting to change it. Firstly, they saw the social and cultural value of the ceremony. Additionally, their relatively privileged social position—marked by higher education, urban residence, stable employment, and/or financial independence—meant that the ceremony did not significantly constrain or harm them, especially in comparison to girls from rural areas. Instead, this privilege enabled them to reinterpret and selectively engage with the tradition, omitting elements that undermined their sense agency, such as partner choice and the timing of marriage.

Thus, the weakened impact of the ceremony shifts the conversation away from viewing it solely as a site of oppression and instead frames it as something negotiable, open to their influence. It becomes a space they can strategically use to their advantage, which brings us to Deniz Kandiyoti's concept of patriarchal bargaining

As I mentioned in the conceptual framework, Kandiyoti's description of patriarchal bargaining in classical patriarchy is relevant, as Ethiopia exhibits many of the same characteristics. Moreover, Kandiyoti emphasizes that women living under such patriarchal systems often adopt interpersonal strategies to enhance their security—for example, by manipulating relationships with their sons or

husbands, sometimes to the extent that the male patriarch loses authority to his wife (Kandiyoti, 1988).

By the same token, participants did not view *Shimgilina* as a source of power over them, but rather as a strategic tool for navigating cultural expectations and building their own forms of power. For many, the ceremony held cultural and social value that they sought to leverage. It was a way to gain credibility, maintain harmony, and affirm familial authority—reflected in comments like "maintain a good relationship with the in-laws," "honor my mother's authority," and "show respect for my family." Through this engagement, participants were not only conforming to expectations but also asserting influence and strengthening their positions within family and community structures.

In her work, Kandiyoti describes women who adopt veiling as a strategic practice to navigate entry into wage labor—a sphere traditionally dominated by men and often considered inappropriate or transgressive for women. By adhering to expected gender norms through veiling, these women mitigate social backlash and maintain respectability, even as they enter spaces typically restricted to men (Kandiyoti, 1988). Similarly, the participants in this study engage with tradition not as an act of submission, but as a strategic effort to balance their "progressive" or "modern" feminist identity with cultural belonging. Their feminist consciousness, education, and professional status—which distinguish them from previous female generations—may have created a sense of detachment from traditional expectations. Participating in *Shimgilina*, then, becomes a way of giving back and reaffirming familial and cultural ties, preserving the tradition as a resource they can draw upon when needed.

For example, Mare, who described herself as a strong-willed and independent person, shared that she had spent years studying in another city and often went against her family's expectations. "I've always been rebellious," she said. "I studied in a different city and used to do the opposite of what they told me. But with this [the ceremony], I wanted to make them happy."

These participants, whether due to their personalities, education, or professional status, do not fit the mold of the "typical" or "traditional" Ethiopian woman. As a result, they often feel a desire to give back and make their families happy by participating in the ceremony, which holds significant cultural value within society.

"I have always struggled with it. You're saying you're possessing the woman, that she doesn't have the right to decide for herself. There's an attempt to rebrand or reimagine it, but it's still the man who has the agency here. It's not easy to reconcile that with modernity, which sees everyone as equal—men and women alike. So, the underlying tone has always made me uncomfortable." (Petros, 41)

Nonetheless, this dynamic should not be seen merely as something they exploit for personal or social gain, such as reading would oversimplify the complexity of their choices.

There is a clear tension here: while these feminists strategically engage with tradition, they also make compromises. It's a form of delayed gratification, where participation often requires the temporary silencing of feminist critiques or their "modern" identities in exchange for forming "respectable" relationships and affirming a sense of cultural belonging. For some, it is easier; for others, it was a struggle.

### 6.1.4 Navigating Feminist Agency through Cultural Attachment

At this point of the analysis, I believe that applying Nego-feminism is crucial for understanding how participants navigate negotiation within the cultural framework in which they are embedded. This framework emphasizes strategies of give-and-take, compromise, and relational balance, offering a lens through which to interpret their actions and choices (Nnaemeka 2004)

For this group, temporarily setting aside their gender consciousness and critiques of *Shimgilina* made strategic sense. Allowing their parents to take control of the ceremony was not a sign of submission, but a calculated trade-off. In exchange for suppressing their discomfort, they gained tangible social benefits: the formal legitimacy of their union—"by doing it the official way," the affirmation of love and respect—"if a man loves you, he's supposed to do that," and continued belonging within their families and communities—"to make my parents happy." Their relationship with their parents, already marked by autonomy made this trade-off feel possible. Because they had made independent choices about their partners and the decision to marry, they felt in a position to give back—to honor their families through Shimgilina without feeling coerced.

Additionally, I believe that understanding what participants negotiate through this process requires careful consideration of their indigenous cultural foundations. As Nnaemeka (2004) argues, meaningful progress—whether in development or gender justice—must build on local knowledge systems rather than override them. (Nnaemeka 2004). This perspective is crucial, as it encourages us to recognize that the value participants place on *Shimgilina* is deeply rooted in cultural traditions and indigenous ways of knowing and being. When they refer to the ceremony as "our way of living," "that is how I was raised," "our way," or "our culture," with pride, they are not merely following

rules—they are actively affirming their cultural identity. Additionally, their understanding of concepts such as "marriage," "respect," "love," and "being official with a loved one" is shaped by a cultural logic that informs who they are and how they relate to others. Recognizing and respecting this cultural grounding allows us to see that their participation is not passive acceptance, but a deliberate effort to reconcile their cultural identity that they embody with their feminist values.

Now that this foundation has been established, Saba Mahmood's conceptualization of agency can be introduced to further deepen the analysis of women's actions. Building on Foucault's concept of subjectivation, Mahmood challenges liberal notions of agency that define it solely as resistance or the pursuit of freedom from oppression. Instead, she argues that power should not be seen only as repressive, but also as productive—shaping what individuals value, desire, and perceive as a meaningful life. Within this framework, agency is understood as a capacity for action that is shaped by, rather than opposed to, structures of subordination. As such, agency may be expressed through practices that sustain continuity, uphold tradition, and maintain social stability, rather than seeking radical transformation (Mahmood 2001).

For the urban, middle-class, educated Ethiopian in this study, desire is shaped not only by feminist consciousness but also by deep cultural attachments and ethical responsibilities within their culture. Being raised in Ethiopia plays a significant role in how they understand key social concepts such as marriage, respect, and pride—particularly as they are expressed through the emotional and symbolic structure of *Shimgilina*.

Although many participants were fully aware of the patriarchal aspects of the ceremony, they did not articulate a desire to challenge or alter it. This was not out of submission or lack of awareness, but rather because their central aspirations were shaped by a desire to carry out the marriage process in a culturally appropriate way—one that introduced families formally, honored their parents, and helped build harmonious relationships with in-laws. These actions were viewed as essential steps toward becoming "respectable women" and "good daughters," in line with societal expectations, but without compromising their personal autonomy, which they believed they already possessed.

For this group, participation in the ritual itself was intentional, as they believed they had already exercised agency in important areas—most notably in selecting their partners and determining the timing of the ceremony. As a result, delegating authority to their parents and preserving gendered elements of the ritual, such as having male elders speak on behalf of the groom, was not experienced as a primary concern or site of struggle. Instead, these aspects were seen as formalities—secondary to the more meaningful decisions that had already affirmed their autonomy and secured their place as a "wife" within both their families and extended communities.

In this sense, the participants' agency did not lie in rejecting or resisting tradition. Their desire for autonomy had already been fulfilled. However, their desire to maintain close family ties remained unmet and was deeply rooted in how they understood marriage, themselves, and their relationships with others. As a result, they chose continuity and decided to hold the ceremony as a way to stay connected to their families and communities.

## **6.2** Reformers: Feminist Shifts in Ceremony

"Even though it wasn't common for women to sit in the negotiation, I wanted my mother and grandmother to be part of it. So, I asked my mom, and she said yes. That's how they became involved in the process." (Tigist, 28)

For this group, Shimgilina was not simply a tradition to be followed, it was a cultural practice they could actively reform. Rather than rejecting the ceremony outright, they sought to reshape it in ways that aligned more closely with their feminist values. They approached it as something through which to exercise agency—meaningful and worth engaging with critically, adapting it where possible to reflect their own beliefs and priorities.

#### **6.2.1** Feminist Critique by Reformers

"It's only men who are sent and who sit to ask the questions" (Tigsit, 28)

Like the other group, these participants are young, middle-class, and well-educated Ethiopians working in the fields of gender and development. However, all members of this group are women. Their backgrounds have given them the tools—both in experience and language—to critically reflect on Shimgilina and clearly articulate the aspects they found oppressive.

For example, Tigist, a 28-year-old with a master's degree in Gender Studies from Addis Ababa University, noted, "It's only men who are sent and who sit to ask the questions," highlighting the absence and lack of representation of women throughout the process. With her background in gender studies and herself being a woman, she emphasized the gender category, paying close attention to the lack of representation of women.

Similarly, Tsige, a 28-year-old currently pursuing her master's degree at Addis Ababa University, spoke with visible frustration when reflecting on the gender dynamics of Shimgilina. "Women

don't have a formal role. Your father and uncle are seen as more important than your mother and aunt—even though they're the ones closest to you," she explained. Her critique was not only analytical but deeply personal, rooted in her experience as a woman. She expressed discomfort with the idea that, in the future, she too might be excluded from her children's ceremonies, despite the emotional labor she would invest in their lives. I believe her frustration was compounded by her own experience: when she tried to include women in her side of the ceremony, her request was denied, with her family dismissively telling her, "Feminism doesn't work here."

These reflections reveal that the ceremony is not accepted uncritically. Participants often problematize the absence of women, drawing on their personal experiences and educational backgrounds.

#### 6.2.2 Shimgilina as Protection, Belonging, and Identity

"My relationship with my partner is not traditional—we live together and are still not married. So, I felt the need to give something back to my parents, to make them feel happy and respected. That's why I agreed to have the ceremony." (Soliyana, 34)

Despite their critiques, the participants also acknowledged the cultural and social significance of the ceremony, as well as its potential protective function for women within the community.

Even Melat, a development practitioner who had her own "type" of *Shimgilina* ceremony two years ago and was among the most vocal in resisting it, acknowledged that the ritual could serve a protective purpose in some contexts.

Melat's connection to the topic runs deep. She has worked extensively on early marriage prevention, she expressed strong frustration with the harmful consequences of such practices,

particularly for young girls. That frustration became personal when she tried to shift the narrative to her sister's ceremony. Melat had hoped to avoid the typical question—"Can he provide for her?"—which she saw as both classist and patriarchal. Despite her efforts, the question was still asked, which left her feeling disappointed and disempowered. By the time it was her turn, Melat took a different approach: rather than hosting a full ceremony, she simply introduced her partner to her parents and allowed them to ask a few questions. It was her way of maintaining a sense of respect and family connection while resisting the more rigid and symbolic performances of tradition.

However, Melat still saw the protective aspect of the shimgilina, as she explained: "It can be used to make sure he doesn't already have another wife or family—it helps prevent scammers," she explained. "If he's trying to marry more than once, the elders won't go along with it." Others described the ceremony as acting like "a third eye"—a means of spotting red flags—and emphasized that it creates a form of communal oversight in which "stakeholders are there to assess whether he's worth it."

Tigist, who was highly critical of the couple's exclusion from the process and opposed the notion that it is the man who solely "decides" to marry the woman, nevertheless acknowledged the ceremony's protective dimension. As she noted, "It gives a sense of who they are sending their daughter to, and the process helps protect her"

Fatima, a 30-year-old feminist from a law background, noted that even the *shimagilewoch* (elders) who are sent can serve as mediators in times of conflict, stating that "they can help resolve issues when something happens." Tsige echoed this sentiment, explaining that "the shimagilewoch are responsible for overseeing the marriage and holding the groom accountable if he does something inappropriate."

It is important to recognize the complex ways in which participants made sense of *Shimgilina*. While they were critical of its patriarchal structure, particularly the marginalization of women in decision-making, they also saw the ceremony as serving a protective function. This perspective can be linked to the cultural respect afforded to elders, especially older men, in Ethiopian society. The term *Shimagilé* is associated with a knowledgeable and experienced person who commands respect. As a result, the involvement of such figures in the ceremony can offer a sense of protection to women (Kassas, and Tessema, 2019), especially in a country that ranks 173rd out of 188 on the Global Gender Inequality Index, reflecting significant disparities in areas such as reproductive health, economic participation, and women's autonomy (Wright, 2020). In many cases, women remain economically and socially dependent on men. Within this context, a tradition such as *Shimgilina*—despite its patriarchal foundations—can function as a form of social protection. As such, these participants did not dismiss outright or abolish without critically considering the structural gaps it may, in certain contexts, be compensating for.

Additionally, like the previous group, these participants emphasized the social and cultural significance of the ceremony, at times, framing it as an expression of identity. They viewed

*Shimgilina* as a means of affirming belonging, strengthening meaningful relationships, and gaining respect within new familial and social networks.

For example, Tigist—despite her frustration with the marginalization of women in the process, including in her own experience—also embraces her Ethiopian identity and the cultural meanings embedded in the ceremony. Reflecting on its significance, she shared: "As an Ethiopian, it's a source of pride for the family. It's part of adulthood and integration into the social fabric.

Soliyana, a 34-year-old woman who identifies as a radical feminist and is deeply committed to challenging gender norms, also saw *Shimgilina* as a way to express care and respect for her family. Her feminism, she explained, was shaped by her father—a man she affectionately described as "a girl's dad" who raised her to be strong and outspoken. Wanting to honor that relationship, she chose to participate in the ceremony not for its traditional meaning, but as a gesture of love. "I did it for my parents," she said. After the ceremony, she could tell how genuinely happy her father was. The next day, during her PhD graduation ceremony, her father proudly introduced her fiancé to relatives, smiling with pride because, as Soliyana put it, "he had already done the ceremony." For her, the decision to take part in Shimgilina was not about complying with patriarchy, but as a way of giving back to a father who had always supported her feminist journey

Tsige, who identifies as a radical feminist but has recently reflected on the need to moderate her stance in certain contexts, saw the ceremony as an opportunity to manage perceptions. She explained: "Especially from my partner's side, they might think I'm easy or not serious if I don't have the ceremony. They wouldn't show me the same level of respect."

For this group, there was a clear awareness of the patriarchal dimensions embedded in the *Shimgilina* ceremony. Yet, participants also engaged with it in culturally meaningful and relational ways. Rather than viewing the ritual solely as a patriarchal imposition, many saw it as a practice that could serve multiple purposes within their social world. Some participants interpreted *Shimgilina* as a form of protection, drawing on the cultural value placed on elders in Ethiopian society. Others, like Soliyana, viewed it as a way to maintain strong familial ties, particularly with parents Additionally, the ceremony functioned as a strategic tool to manage perceptions and affirm their social legitimacy as respectable women.

In their own culturally grounded ways, participants used *Shimgilina* to strengthen interpersonal relationships—honoring elders, reinforcing family bonds, and performing respectability in a way that aligned with their Ethiopian identity. These interpretations reflected a deep understanding of concepts such as respect, belonging, and interdependence, shaped by their social context rather than external feminist ideals alone.

## 6.2.3 Shaping the Ritual: Autonomy Through Cultural Influence

This group consists of highly educated women, all of whom married between the ages of 27 and 30 and entered their relationships through personal choice rather than family arrangement. They independently selected their partners—often beginning as boyfriend–girlfriend relationships—with no parental involvement in the matchmaking process. Importantly, the timing and initiation of the *Shimgilina* ceremony were determined by the couple themselves, reflecting a greater degree of autonomy. One illustrative example involves a participant who was in a relationship with a non-Ethiopian partner and chose to formalize the union on her terms, based on her priorities.

This marks a significant difference between the participants in my study and girls under the age of 18, who are often married off by their parents and face a heightened risk of spousal abuse and domestic violence throughout their marriages (Abera et al. 2020). The women in my study entered marriage based on partnership or love, whereas in many rural cases, girls are married either to prevent premarital pregnancy(Kok et al. 2023) or to strengthen familial ties (Abera et al. 2020).

However, despite their clear exercise of personal autonomy in partner selection and their feminist critiques of women's marginalization, patriarchal elements remained embedded in most of their *Shimgilina* ceremonies. This group was not homogenous—they navigated different cultural expectations, family dynamics, and feminist questions—but across their varied experiences, certain patterns persisted. In nearly all cases, it was the groom who initiated the ceremony by "sending the elders," a role always occupied by men.

Only one participant deviated from this norm, opting instead for an informal dinner with her now-husband and her parents. Even in this instance of resistance—such as Melat's—traditional gendered expectations still surfaced. All of the other participants, including Melat, recalled the question being asked by the elders from their side, "Can he provide for her?" In Melat's case, although she actively resisted many traditional elements, a similar expectation emerged in a different form: when she introduced him to her family, they asked, "What do you do?" However, when she met his family, they did not ask her the same question. This subtle omission reinforces how deeply ingrained assumptions about male provision and female dependency continue to shape interactions, even in less conventional or intentionally subversive settings.

As an analytical distinction, I argue that the first group, the Reinterpreters, believed that they possessed a form of agency rooted in their autonomy to make choices such as relocating to new cities, independently choosing their partners, and, in many cases, living with them without parental involvement. Even those who did not cohabit before marriage shared stories of being supported by their families,—"to make these decisions on their own". These experiences contributed to a strong sense of independence, which, in my reading, formed the basis of their sense of autonomy and how they negotiated with the shimgilina ceremony.

In contrast, this group—although some also lived with their partners—did not frame those choices as central expressions of agency. Instead, they saw their agency as more visible in how they engaged with the *Shimgilina* ceremony itself. Their sense of agency emerged through efforts to reshape its structure, procedures, and symbolic meanings.

"I wanted to have it because I wanted to influence the ceremony, you know. I wanted my mom to sit and ask questions." (Tsige, 28)

Still, I would argue that although this group did not explicitly frame their actions prior to the ceremony as feminist or agentic, they were nonetheless exercising a form of agency even before and during the ceremony for two key reasons. First, none of the participants expressed regret or negative feelings about their participation. On the contrary, they described the experience using words like "exciting," "empowering," "fun," and sometimes "normal". Their urban, educated, and middle-class backgrounds appeared to shield them from feelings of constraint, allowing them to engage with the tradition in ways that felt voluntary and affirming. I'm sure that if we asked people who participated in the traditional ceremony, they wouldn't describe it that way.

Second, several participants explained that their decision to participate in *Shimgilina* was driven by a desire to "show respect for their family" or to avoid being perceived as "easy." In this sense, their choice was not rooted in coercion but in a strategic effort to preserve and strengthen familial and social relationships. This reflects what Deniz Kandiyoti (1988) calls patriarchal bargaining—a form of agency in which women navigate patriarchal structures not by overt resistance, but by making calculated decisions that allow them to secure respect, legitimacy, and emotional proximity within existing social norms (Kandiyoti, 1988). For these women, as with the previous group, *Shimgilina* was not experienced as an oppressive practice. Because they are educated, middle-class, urban women, the ceremony was not forced upon them. Instead, it was something they actively chose to participate in—viewing it as an opportunity to build social capital and strengthen familial ties.

That said, for these feminists, simply choosing their partners and nurturing personal relationships using this ceremony wasn't enough to fully express their sense of autonomy. They recognized that the ceremony still carried symbolic and social weight in their lives. As a result, they actively sought to reshape certain aspects of the ritual that they found limiting or misaligned with their values

For instance, Fatima shared that she chose to have the *Shimgilina* ceremony to "acknowledge [her] family" and "allow them to be part of [her] marriage." She defines her feminism as "resisting restrictive structures that hold back women," and for her, a key expression of both her feminism and sense of agency was having the ceremony, was ensured that no one asked about her partner's financial status or discussed dowry.

In her family's ethnic tradition, the ceremony unfolds in two stages: first, the women go to request the bride's hand, followed by the men on another day. Fatima described the first stage as going "perfectly." Her mother and grandmother—both of whom she credits as shaping her feminist values—led the process without inquiring about her partner's finances or social standing. She appreciated that this part of the ceremony allowed her family to ask meaningful questions about her partner's intentions without reducing the interaction to economic terms.

However, the second stage, when the men entered, felt markedly different. As she put it, "They might even end up using the power they think they have," she said. Over forty men from both sides were involved in this phase, and her mother served as a key mediator and communicator on her behalf. Fatima recalled telling her mother, "This isn't about exchanging me or giving me away like property—it's just part of the culture. It's a way for the families to be introduced to one another."

Nonetheless, moments of discomfort persisted. Fatima recalled that during the negotiation, one elder from her side remarked, "Even if he takes additional wives, they will never get divorced." She recognized that such statements reflected the values of those involved in the ceremony rather than those of herself or her partner. "So, I just let it go," she noted. "But at the same time, because of my influence, they didn't bring up dowry or anything like that."

For Tigist, "representation was everything" when it came to the Shimgilina ceremony. She recalled telling her mother, "I want women from my side of the family to sit with the matchmakers, because they also helped raise me." Her mother agreed without hesitation. "She just said, 'Of

course, "Tigist explained. This decision allowed her to include her mother, grandmother, and one of her mother's close friends in the negotiation process. Curious about how her family accepted the idea so easily, I asked her directly how it was easy for her. She smiled and said, "My family is very open-minded." She added, "Both of my parents walked me down the aisle. That's not something people usually do, but it felt right. It was symbolic of the way I was raised."

Still, her vision of equal representation didn't fully carry over to her husband's side. "I wanted him to send women too," she admitted, "but it just got too complicated. Who would he even send?

And would the other men be okay with that? Probably not." Ultimately, she let it go. "To be honest," she added, "most of the men on his side weren't just elders—they were church figures. So, it was hard."

Soliyana, who identifies as a radical feminist and describes herself as someone who wants to "disrupt" conventional norms, initially envisioned an all-women Shimgilina ceremony. "I wanted to flip the script and shock people," she explained. "But in the end, it just wasn't feasible—there were too many factors to consider: logistics, family dynamics, all of it."

Because her partner was from a different country, she had more flexibility in influencing who would be sent on his behalf. Although she initially wanted to include women on his side, it wasn't feasible. "I really tried," she said, "but I couldn't find anyone who could take on the role." Instead, she appointed one of her feminist male friends to serve as an elder—someone she felt genuinely reflected her values. Thus, when the customary question came up—how her partner would provide

for her—her chosen representative confidently responded, "She's very capable and can take care of herself." Soliyana recalled, "I actually found it very empowering."

Drawing on Obioma Nnaemeka's (2004) concept of Nego-feminism—a feminism grounded in negotiation, give-and-take, and no-ego—these participants did not reject tradition outright. Instead, they exercised agency by engaging with *Shimgilina* as a site of relational negotiation. Their stories reflect the core principles of Nego-feminism: flexibility, mutual exchange, and the importance of lived experience. Moreover, in line with Nnaemeka's concept, they did not position culture as something to resist but rather worked within it, viewing their feminist development as emerging from—not against—their cultural identity (Nnaemeka 2004). For this group, there was a greater sense of tension—each participant navigated a complex mix of enabling and limiting factors.

For Fatima, her agency was reinforced by the presence of feminist role models—her mother and grandmother—as well as by an ethnic tradition that allowed for a women-led version of *Shimgilina*. She exercised influence by ensuring that discussions around dowry and financial status were excluded from the ceremony. However, her ability to shape the entire ritual was limited. During the male-dominated stage—where over forty men participated and where she was not present—conversations unfolded that she could not control, many of which reflected patriarchal and religious norms, including the acceptance of polygamy. Aware of her position as a young woman, she strategically relied on her mother, who held greater social standing, to advocate on her behalf. Still, given the broader power structures at play, she chose to let certain things go. This decision was not a surrender but a calculated act of balance—an acknowledgment that while she could influence specific elements, transforming the entire system was neither feasible nor necessary to

assert her agency. Even within the constraints of the ceremony, she was able to negotiate and gain something on her terms.

Tigist similarly drew strength from her family's open-mindedness. Her feminist vision of including maternal figures in the negotiation process was realized on her side, reflecting a form of relational agency. However, she encountered limitations when attempting to extend this inclusion to her husband's side, where religious authority and rigid gender roles prevailed. Rather than escalate conflict, Tigist chose to let go of that demand. While this decision may appear as a concession, it reflects the logic of give-and-take. She was able to influence the cultural script to include those who mattered to her—like her mother and grandmother—but recognized that some aspects were beyond her control and chose to strategically let them go.

Nego-feminism is especially valuable for understanding the complexities revealed in these stories because it centers lived experience as a site of feminist theorizing. Rather than viewing feminism as a fixed set of principles or a rigid form of resistance, Nego-feminism invites us to pay attention to how women engage in everyday negotiations—talking to family members, letting certain things go, or asserting influence where they can. These small, strategic acts—what Obioma Nnaemeka calls "the story of now"—are where feminism is practiced (Nnaemeka 2004).

When we examine their lived experiences, we also see the importance of flexibility. Each woman made choices shaped by the space available to her, using the tools at her disposal to assert her values. This is where the flexibility of Nego-feminism becomes crucial—it reminds us that feminism cannot follow a one-size-fits-all model(Nnaemeka 2004). For instance, Tigist could not

confront patriarchy in the same way Fatima did—her context was different. Their families, ethnic culture, religious affiliations, and surrounding social networks shaped what was possible for them. What Nego-feminism offers is a framework that honors these differences, acknowledging that feminist action can look like resistance, compromise, or strategic silence. Through this lens, we can see that each woman was not only navigating culture but also contributing to it—on her own terms.

Across all three stories, agency is exercised not by rejecting Shimgilina, but by working through it—treating culture not as an obstacle, but as a resource for feminist expression. These women do not seek to dismantle the tradition outright; instead, they reconfigure it in subtle yet meaningful ways to reflect their values. As one participant shared, "I wanted my mother and grandmother to be part of the negotiation, because they also helped raise me." This inclusion reflects not a break from tradition, but a reimagining of it—one that honors both cultural heritage and gendered justice.

Their actions are rooted in deeply held cultural values of family pride, respect, and social legitimacy. As another participant explained, "I wanted to influence it" Rather than viewing these norms as barriers, participants saw them as frameworks through which they could assert their feminist commitments. In their view, culture was not the enemy of feminism—it was the ground from which new possibilities could emerge.

By intentionally including women in the negotiation, choosing supportive elders to represent them, or steering conversations away from practices like dowry, they enact a form of agency deeply embedded in cultural and relational logics, in a way that made sense to them individually.

The third aspect is that these women practiced no-ego feminism, as theorized by Obioma Nnaemeka(Nnaemeka 2004). They recognized Shimgilina not just as a personal milestone, but as a communal event requiring negotiation between individual feminist values and collective family expectations. So that is why "they had to let it go" even when others are using "authorities that they think they have" or "family logistics become complicated"

Moreover, Saba Mahmood building on Foucauldian's reading of power reminds us that power is not solely repressive; it is also productive, shaping how individuals come to understand their roles, responsibilities, and emotional investments (Mahmood 2001). For the urban, middle-class, educated Ethiopian women in this group, agency unfolded through the negotiation between their feminist sensibilities and deeply rooted cultural identities.

While they recognized Shimgilina as a patriarchal institution, participants also understood it as a meaningful cultural ritual that reinforced family bonds and facilitated the merging of kin networks. Their desire to participate was not merely an act of compliance; rather, it reflected a culturally informed aspiration to enter marriage with social legitimacy and communal recognition. As one participant put it, "if I don't do it, they might think I am easy."

At the same time, their feminist sensibilities urged them to question the exclusion of women from this union or control the discourse. Their desire to be seen as "respectable women" in the eyes of their in-laws was balanced by a responsibility to advocate for greater gender inclusion. As a result, they asked critical questions, pushed for the inclusion of maternal figures, and subtly reshaped the ceremony's structure. This push for representation was not solely about achieving gender parity;

it reflected a deeper aspiration to cultivate a relational and gender-inclusive community as they transitioned into married life.

This is precisely where their agency lies—not in outright rejection, but in the distinct desires that lead to a negotiation between honoring familial expectations and embodying feminist values.

#### 6.2.4 Conclusion

I identified two feminist approaches to the Shimgilina ceremony among the middle-class, educated, urban Ethiopians. The first group, the *Reinterpreters*, view the ritual as largely symbolic—something they engage in primarily to legitimize their relationships and fulfill social expectations. The second group, the *Reformers*, seeks to reshape the ceremony in ways that better reflect their feminist values. Although their approaches differ, both groups demonstrate forms of agency that are strategic, nuanced, and deeply rooted in cultural context.

By applying the frameworks of intersectionality, Nego-feminism, patriarchal bargaining, and Saba Mahmood's concept of embodied agency, I have shown that the participants are exercising agency within a complex web of cultural expectations, family dynamics, sense of personal autonomy, and feminist commitments. Rather than acting in isolation, their ecisions are shaped by overlapping pressures—such as the desire to honor familial traditions, assert independence, preserve social legitimacy, and remain true to feminist values.

Crucially, the points at which participants locate room for negotiation vary. For some—particularly those I describe as *Reinterpreters*—agency is exercised prior to the ceremony itself,

through the selection of their partner, the decision to cohabit without parental involvement, or the choice to time the ceremony on their own terms. Because they perceive the most meaningful acts of autonomy to have already occurred, they do not seek to influence the ceremony's structure. Instead, they engage with it as a largely symbolic ritual—one that offers social recognition and familial harmony, but no longer constrains them.

Others—the *Reformers*—see the ceremony itself as the primary site of negotiation. For them, agency involves actively reshaping the ritual to include women, avoid patriarchal tropes such as economic vetting, or introduce new forms of representation. Their feminist consciousness motivates them to intervene more directly, treating the ceremony not simply as a formality but as a meaningful space in which gender norms can be challenged, even if only partially.

Together, these divergent strategies reveal that feminist agency is not a uniform act of resistance or acceptance but a situated and relational process. It is exercised differently depending on one's social position, family relationships, and sense of autonomy.

## 7 Conclusion

This thesis set out to explore a central puzzle: How do young, urban, educated, middle-class Ethiopians who identify as feminists engage with *Shimgilina*, a traditionally patriarchal prewedding ceremony? Rather than whether participation in such rituals signals conformity or lack of agency, I asked how these feminists negotiate their feminist values within deeply embedded cultural institutions. At the heart of this inquiry was a desire to understand how agency is exercised not only through resistance or rejection but also through subtle, strategic negotiation within existing structures.

To explore this question, I conducted qualitative research with ten self-identified feminists—nine women and one man—in Addis Ababa who had the *Shimgilina* ceremony. While they shared similar social locations shaped by age, urban residence, higher education, and middle-class status, their perspectives and strategies for engaging with the ritual varied considerably. Drawing on semi-structured interviews, I examined their lived experiences with a focus on the ceremony itself, the decisions they made around it, and the tensions or conflicts that emerged in the process.

Empirically, the study revealed two distinct patterns in how participants navigated their agency. The first group experienced a strong sense of autonomy before the *Shimgilina* ceremony—especially in their ability to choose their life partners independently. This prior exercise of choice gave them a sense of control, which made them less inclined to challenge or intervene in the structure of the ceremony itself. Instead, they allowed their families to take control of the ritual,

often reinterpreting its meaning in personal and symbolic terms—seeing it not as something done *to* them, but as a gesture of respect for their families or cultural continuity.

By contrast, the second group did not feel they had exercised meaningful autonomy prior to the ceremony and instead viewed *Shimgilina* as the critical site through which to express agency. These women actively engaged with the ceremony, seeking to reform it by incorporating feminist values—such as including maternal figures in the negotiation or pushing back against patriarchal assumptions. For them, the ceremony was not a passive tradition but a space of negotiation, transformation, and cultural critique.

By weaving together these four frameworks—intersectionality, patriarchal bargaining, Nego-feminism, and Mahmood's rethinking of agency—this study not only applies existing theories but also extends them by grounding them in the lived experiences of urban, middle-class Ethiopian feminists navigating a deeply gendered cultural ritual.

While Kimberlé Crenshaw introduced intersectionality within a legal framework to expose how overlapping systems of oppression are often overlooked, this study broadens the concept by applying it to cultural identity and lived experience. Rather than viewing national identity as a fixed structural category, I analyze how participants actively embody and express their Ethiopian identity through rituals like Shimgilina. This identity is not solely as sources of constraint, but also a frameworks of meaning and potential empowerment. In doing so, I extend Crenshaw's framework to show that intersecting identities can produce not only compounded oppression but also culturally grounded forms of agency and resistance.

This study extends the concept of patriarchal bargaining by emphasizing that interpersonal strategies are not uniform but deeply shaped by cultural meanings and relational dynamics. The "bargain" is highly flexible and context-specific, influenced not only by the type of patriarchy present but also by historical, cultural, and socio-political factors. Identities such as the "good daughter," "good wife," or "respectable in-law" are not universal categories—they are negotiated within specific cultural frameworks that give them meaning and value.

This study strengthens Nego-feminism by providing empirical evidence of how African women negotiate cultural rituals in relational, context-specific ways. It shows that even among similarly positioned women, feminist negotiation takes diverse forms—some assert agency through choosing their partners, while others reshape the ritual itself. This supports Nnaemeka's emphasis on flexibility and expands it by demonstrating that negotiation happens at different stages, through various strategies, and for different purposes. Furthermore, while Nego-feminism values indigenous cultural knowledge, it often lacks concrete illustrations of how culture can serve as a site of feminist theorizing. This research bridges that gap by showing how participants drew from cultural logics—like familial pride, community belonging, and respect—not only as constraints but as tools for expressing feminist values. Culture, then, becomes not just something to negotiate within, but a generative source of feminist knowledge.

This study also adds depth to Saba Mahmood's rethinking of agency. While Mahmood challenges liberal feminist frameworks by showing how women can exercise agency through tradition, she tends to present groups—like the pious women in Cairo—as relatively unified in their orientation. In contrast, my findings reveal that even within a seemingly homogenous group, experiences of

autonomy and desire vary significantly. One of the most notable points of divergence among participants was their sense of autonomy.

These factors influenced not only their motivations but also how they navigated the same cultural ritual in different ways. This suggests that Mahmood's "network of concepts"—desire, responsibility, and affect—is not fixed, but fluid and contingent on one's position within intersecting power structures. Ironically, while agency is deeply rooted in cultural norms and practices, it is also highly individualized, revealing the relational and context-specific nature of how power operates in everyday life.

# **Bibliography**

- Abdurahman, Dureti, Nega Assefa, and Yemane Berhane. 2022. "Parents' Intention toward Early Marriage of Their Adolescent Girls in Eastern Ethiopia: A Community-Based Cross-Sectional Study from a Social Norms Perspective." *Frontiers in Global Women's Health* 3 (October):911648. https://doi.org/10.3389/fgwh.2022.911648.
- Abera, Mikyas, Ansha Nega, Yifokire Tefera, and Abebaw Addis Gelagay. 2020. "Early Marriage and Women's Empowerment: The Case of Child-Brides in Amhara National Regional State, Ethiopia." *BMC International Health and Human Rights* 20 (1): 30. https://doi.org/10.1186/s12914-020-00249-5.
- Adamu, Abebaw Y. 2013. "Diversity in Ethiopia." *The International Journal of Community Diversity* 12 (3): 17–27. https://doi.org/10.18848/2327-0004/CGP/v12i03/39924.
- Adeoye-Olatunde, Omolola A., and Nicole L. Olenik. 2021. "Research and Scholarly Methods: Semi-structured Interviews." *JACCP: JOURNAL OF THE AMERICAN COLLEGE OF CLINICAL PHARMACY* 4 (10): 1358–67. https://doi.org/10.1002/jac5.1441.
- Alwedinani, Dr. Jawaher. 2017. "Bargaining With Patriarchy: Women's Subject Choices and Patriarchal Marriage Norms." International Journal of Gender & Women's Studies 5 (2). https://doi.org/10.15640/ijgws.v5n2a2.
- Augustyniak, Zuzanna. 2009. Marriages in Ethiopia. Studies of the Department of African Languages and Cultures 43. Warsaw: University of Warsaw.
- Baker, Maureen and Vivienne Elizabeth.2013. "Did you just ask me to marry you?": The gendered nature of heterosexual relationship progressions" Women's Studies Journal Volume 27(2) 32-43
- Balew Liyew, Estifanos. 2024. "Marriage Practice: A Comparative Analysis between Chinese and Ethiopian People." *Community and Interculturality in Dialogue* 4 (August):103. https://doi.org/10.56294/cid2024103.
- Beyene, Gemechu. 2006. "Marriage Practices Among The Gidda Oromo, Northern Wollega, Ethiopia."
- Birhan, Aemro. 2024. "Ethnic Identity and National Unity in Ethiopia: Challenges and Opportunities for Building National Consensus." *Journal of Political Science and International Relations* 7 (2): 40–47. https://doi.org/10.11648/j.jpsir.20240702.12.
- Bukamal, Hanin. 2022. "Deconstructing Insider—Outsider Researcher Positionality." *British Journal of Special Education* 49 (3): 327–49. https://doi.org/10.1111/1467-8578.12426.
- Byrne, Deirdre. 2020. "Decolonial African Feminism for White Allies" 21 (7)
- Carter, Julia. 2022. "Traditional Inequalities and Inequalities of Tradition: Gender, Weddings, and Whiteness." *Sociological Research Online* 27 (1): 60–76. https://doi.org/10.1177/1360780421990021.
- Cherinet, Haregewoin, and Emebet Mulugeta. 2003. "A Profile on Gender Relations Towards Gender Equality in Ethiopia." Report. Sida.
- Crenshaw, Kimberle. 1989. "Demarginalizing the Intersection of Race and Sex: A Black Feminist Critique of Antidiscrimination Doctrine, Feminist Theory and Antiracist Politics." University of Chicago Legal Forum: Vol. 1989: Iss. 1, Article 8

- Crenshaw, Kimberlé. 1991. "Mapping the Margins: Intersectionality, Identity Politics, and Violence against Women of Color." Stanford Law Review 43 (6): 1241–99. https://doi.org/10.2307/1229039.
- Dessie, Anteneh Mengist, Denekew Tenaw Anley, Melkamu Aderajew Zemene, Natnael Atnafu Gebeyehu, Getachew Asmare Adella, Gizachew Ambaw Kassie, Misganaw Asmamaw Mengstie, et al. 2023. "Trend, Determinants, and Future Prospect of Child Marriage in the Amhara Region, Ethiopia: A Multivariate Decomposition Analysis." *Frontiers in Public Health* 11 (September):1132148. https://doi.org/10.3389/fpubh.2023.1132148.
- Dessie, Enchalew, and Kebede Lemu Bekelcha. 2023. "Marriage Practice and Its Type among Motta Community in East Gojjam Zone," no. 2583.
- Dewandaru, Sanastri Nurdityaning, and Endah Triastuti. 2021. "Gender Role Commodification by the Wedding Industry:" In Jakarta, Indonesia. https://doi.org/10.2991/assehr.k.210531.028.
- Dorosh, Paul, and Emily Schmidt. 2010. "The Rural-Urban Transformation in Ethiopia."
- Ellingsæter, Anne Lise. 2023. "Resisting or Maintaining Gender Inequality? Wedding Traditions among Norwegian Millennials." *Acta Sociologica* 66 (1): 44–58. https://doi.org/10.1177/00016993221074826.
- Erulkar, Annabel. 2013. "Early Marriage, Marital Relations and Intimate Partner Violence in Ethiopia." *International Perspectives on Sexual and Reproductive Health* 39 (01): 006–013. https://doi.org/10.1363/3900613.
- Erulkar, Annabel. 2022. "Changes in the Prevalence of Child Marriage in Ethiopia, 2005–2016." *Reproductive Health* 19 (S1): 195. https://doi.org/10.1186/s12978-021-01234-4.
- Fairchild, Emily. 2014. "Examining Wedding Rituals through a Multidimensional Gender Lens: The Analytic Importance of Attending to (In)Consistency." *Journal of Contemporary Ethnography* 43 (3): 361–89. https://doi.org/10.1177/0891241613497750.
- Froschauer, Ursula, and Kevin Durrheim. 2019. "'It's the Bride's Day': The Paradox of Women's Emancipation." *Feminism & Psychology* 29 (1): 58–75. https://doi.org/10.1177/0959353518803975.
- Gatwiri, Glory Joy, and Anne Mumbi. 2016. "Silence as Power: Women Bargaining With Patriarchy in Kenya." Social Alternatives. Vol. 35.
- Gebeyehu, Natnael Atnafu, Molalign Melese Gesese, Kirubel Dagnaw Tegegne, Yenalem Solomon Kebede, Gizachew Ambaw Kassie, Misganaw Asmamaw Mengstie, Melkamu Aderajaw Zemene, et al. 2023. "Early Marriage and Its Associated Factors among Women in Ethiopia: Systematic Reviews and Meta-Analysis." Edited by Takele Gezahegn Demie. *PLOS ONE* 18 (11): e0292625. https://doi.org/10.1371/journal.pone.0292625.
- Gebrekirstoskassa, Serkalem.2018. "THE IMPACT AND INFLUENCE OF GLOBALIZATION ON ETHIOPIAN CULTURE: HYBRIDIZATION OF AMHARIC WITH ENGLISH IN THE YOUTH CULTURE OF ADDIS ABABA."
- Hailu, Yerasework Kebede. 2021. "Did Ethiopia Survive Coloniality: Eurocentric Colonial Interpretations of Ethiopian History; Struggle against Territorial Colonialism; The Three Unique Historical Aspects of Ethiopia; Challenges of Modernization; Did Ethiopia Avoid Colonialism." *Journal of Decolonising Disciplines* 2 (2). https://doi.org/10.35293/jdd.v2i2.18.
- Harding, Sandra. 1987. Feminism & Methodology: Social Science Issues. Indiana University Press.
- Helma Lutz. 2015. "Intersectionality as Method." *DiGeSt. Journal of Diversity and Gender Studies* 2 (1–2): 39. https://doi.org/10.11116/jdivegendstud.2.1-2.0039.

- Jackson, Kelly F., Sara Goodkind, Mery Diaz, Sharvari Karandikar, Ramona Beltrán, Mimi E. Kim, Jennifer R. Zelnick, et al. 2024. "Positionality in Critical Feminist Scholarship: Situating Social Locations and Power Within Knowledge Production." *Affilia* 39 (1): 5–11. https://doi.org/10.1177/08861099231219848.
- John, Neetu A., Jeffrey Edmeades, Lydia Murithi, and Iman Barre. 2019. "Child Marriage and Relationship Quality in Ethiopia." *Culture, Health & Sexuality* 21 (8): 853–66. https://doi.org/10.1080/13691058.2018.1520919.
- Kandiyoti, Deniz.1988. "Bargaining with Patriarchy."
- Kassas, Ebabushe and Meseret Tessema. 2019. "An Overview of Ageing and Culture in Ethiopian Context" Journal of Advances and Scholarly Researches in Allied Education Vol. 16 (1): 53-57
- Kelecha, Yirgalem Tola, Amanuel Albene Ayele, Habtamu Samuel Goda, Mesarch Hailu Demissie, and Temesgen Mohammed Toma. 2024. "Inequitable Gender Norms and Its Associated Factors among University Students in Southern Ethiopia: A Cross-Sectional Study, 2022." *Frontiers in Public Health* 12 (December):1462782. https://doi.org/10.3389/fpubh.2024.1462782.
- Kifle, Dereje. 2013. "An Ethnographic Study of Events and Their Symbolic Representations in a Wedding Ceremony of Ethiopian Orthodox Church: A Bird's Eye View."
- King, Ian W., and Annick Schramme, eds. 2019. *Cultural Governance in a Global Context: An International Perspective on Art Organizations*. Palgrave Studies in Business, Arts and Humanities. Cham: Springer International Publishing. https://doi.org/10.1007/978-3-319-98860-3.
- Kok, Maryse C., Tasneem Kakal, Abeje Berhanu Kassegne, Irwan M. Hidayana, Alister Munthali, J. Anitha Menon, Paulo Pires, Tabither Gitau, and Anke Van Der Kwaak. 2023. "Drivers of Child Marriage in Specific Settings of Ethiopia, Indonesia, Kenya, Malawi, Mozambique and Zambia Findings from the Yes I Do! Baseline Study." *BMC Public Health* 23 (1): 794. https://doi.org/10.1186/s12889-023-15697-6.
- Mahmood, Saba. 2001. "Feminist Theory, Embodiment, and the Docile Agent: Some Reflections on the Egyptian Islamic Revival." *Cultural Anthropology* 16 (2): 202–36. https://doi.org/10.1525/can.2001.16.2.202.
- Mengesha, Astair Gm, Solomon Deressa, and Yeshi Emabet Imagnu. 1996. "The Context of Ethiopian Weddings in North America: Ritualized Communication." *Northeast African Studies* 3 (3): 111–25. https://doi.org/10.1353/nas.1996.0019.
- Merriam, Sharan B., Juanita Johnson-Bailey, Ming-Yeh Lee, Youngwha Kee, Gabo Ntseane, and Mazanah Muhamad. 2001. "Power and Positionality: Negotiating Insider/Outsider Status within and across Cultures." *International Journal of Lifelong Education* 20 (5): 405–16. https://doi.org/10.1080/02601370120490.
- Musingafi, Maxwell Constantine Chando. 2023. "Conceptualising Feminism in Africa:" In Advances in Religious and Cultural Studies, edited by Maxwell Constantine Chando Musingafi and Chipo Hungwe, 1–19. IGI Global. https://doi.org/10.4018/978-1-6684-9721-0.ch001.
- Naderifar, Mahin, Hamideh Goli, and Fereshteh Ghaljaie. 2017. "Snowball Sampling: A Purposeful Method of Sampling in Qualitative Research." *Strides in Development of Medical Education* 14 (3). https://doi.org/10.5812/sdme.67670.

- Nnaemeka, Obioma. 2004. "Nego-Feminism: Theorizing, Practicing, and Pruning Africa's Way." *Signs: Journal of Women in Culture and Society* 29 (2): 357–85. https://doi.org/10.1086/378553.
- Nowell, Lorelli S., Jill M. Norris, Deborah E. White, and Nancy J. Moules. 2017. "Thematic Analysis: Striving to Meet the Trustworthiness Criteria." *International Journal of Qualitative Methods* 16 (1): 1609406917733847. https://doi.org/10.1177/1609406917733847.
- Orb, Angelica, Laurel Eisenhauer, and Dianne Wynaden. 2001. "Ethics in Qualitative Research." *Journal of Nursing Scholarship* 33 (1): 93–96. https://doi.org/10.1111/j.1547-5069.2001.00093.x.
- Opara, Chioma. 2024. "Integrative Dynamics of Femalism in African Feminist Discourse." Application/pdf, 161858 b. https://doi.org/10.25365/PHAIDRA.586 04.
- Pratelli, Nicole. 2019. "UC 031 311 342 Bezeichnung Des Bachelorstudiums."
- Sande Lie, Jon Harald, and Berouk Mesin. 2018. "Ethiopia: A Political Economy Analysis." Norwegian Institute of International Afairs.
- Schweingruber, David, Sine Anahita, and Nancy Berns. 2004. "'Popping the Question' When the Answer Is Known: The Engagement Proposal as Performance." *Sociological Focus* 37 (2): 143–61. https://doi.org/10.1080/00380237.2004.10571239.
- Tătar, Corina Florina, Seedou Mukthar Sonko, Alphonse Sambou, Isobel Manuel, Workie Wuletaw, and Marius Ioan Tătar. 2024. "BETWEEN TRADITION AND MODERNITY: ETHNOGRAPHIC PERSPECTIVES ON MARRIAGE PRACTICES IN ETHIOPIA, NAMIBIA, ROMANIA AND SENEGAL." GeoJournal of Tourism and Geosites 57 (4 supplement): 2029–40. https://doi.org/10.30892/gtg.574spl17-1370.
- Telila, Shambel Tufa.2020. "Socio-Economic and Cultural Values of Marriage among of Arsi Oromos, in Hethosa Woreda in Oromia Regional State, Ethiopia."
- Worku Kalore. 2019. "Social Structure, Kinship and Death Rituals among the Hadiya." ResearchGate.
  - August27,2019.https://www.researchgate.net/publication/335611374\_Social\_structure\_kinship\_and\_death\_rituals\_among\_the\_Hadiya.
- Wright, Ain.2020. "Closing the Gender Gap: Women's Rights in Ethiopia and Mexico."